

The Different Faces of AI Ethics Across the World: A Principle-Implementation Gap Analysis

Lionel Nganyewou Tidjon, Foutse Khomh

▶ To cite this version:

Lionel Nganyewou Tidjon, Foutse Khomh. The Different Faces of AI Ethics Across the World: A Principle-Implementation Gap Analysis. 2022. hal-03689128

HAL Id: hal-03689128 https://hal.science/hal-03689128

Preprint submitted on 7 Jun 2022

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.

The Different Faces of AI Ethics Across the World: A Principle-Implementation Gap Analysis

Lionel Tidjon and Foutse Khomh, Senior Member, IEEE

Abstract—Artificial Intelligence (AI) is transforming our daily life with several applications in healthcare, space exploration, banking and finance. These rapid progresses in AI have brought increasing attention to the potential impacts of AI technologies on society, with ethically questionable consequences. In recent years, several ethical principles have been released by governments, national and international organisations. These principles outline high-level precepts to guide the ethical development, deployment, and governance of AI. However, the abstract nature, diversity, and context-dependency of these principles make them difficult to implement and operationalize, resulting in gaps between principles and their execution. Most recent work analysed and summarized existing AI principles and guidelines but they did not provide findings on principle-implementation gaps and how to mitigate them. These findings are particularly important to ensure that AI implementations are aligned with ethical principles and values. In this paper, we provide a contextual and global evaluation of current ethical AI principles for all continents, with the aim to identify potential principle characteristics tailored to specific countries or applicable across countries. Next, we analyze the current level of AI readiness and current implementations of ethical AI principles in different countries, to identify gaps in the implementation of AI principles and their causes. Finally, we propose recommendations to mitigate the principle-implementation gaps.

Index Terms—AI, Ethics, Ethical Principles, Principle Implementation, Gap Analysis, EthicsOps.

1 INTRODUCTION

AI achieved successful results in several domains such as healthcare [1], automotive [2], and aviation [3]. The powerful and profound impact of AI on society led to several debates about the principles and values that must guide its implementation and use. In recent years, several ethical AI principles have been released by governments, national, and international organisations. Some of these principles are generic while others are tailored to a specific environment or context. Most generic principles have been designed by international organisations, with the aim to provide abstract and inclusive principles, that will lead the implementation and use of AI in member countries or companies. Contextual principles have been designed by national organizations and governments to take into account the specific AI factors in a given country. For example, no robotics principle is provided for a country with no robotics industry. As a consequence, countries around the world are having different principles tailored to their context in addition to generic principles from the international organizations to which they are affiliated. This diversity and context-dependency of ethical AI principles make them difficult to implement and operationalize; resulting in gaps between the stated principles and their execution.

Recently, researchers have analysed and summarized AI principles proposed on some continents [4]–[7]. However, to the best of our knowledge, none of these studies examined the gaps between the principles and their implementation. This paper aims to fill this gap in the literature by analyzing ethical AI principles around the world and the possible gaps

between the principles and their implementations. We also investigate the root causes of principle-implementation gaps and propose mitigation strategies. The analysis of ethical AI principles around the world is important for understanding the characteristics of ethical principles tailored to specific countries, and identifying similarities between principles from different countries, as well as similarities between global principles from different continents and those from international organizations such as the Organization for Economic Cooperation and Development (OECD), the Global Partnership on AI (GPAI), and the Group of Twenty (G20).

In this paper, we provide a contextual and holistic evaluation of current ethical AI principles and their implementations by analyzing a set of 100 documents and websites containing 100 AI principles from 29 countries in the 6 continents and 121 implementations of these principles from 14 countries in the 6 continents (e.g., guidelines, standards, tools). The documents are selected based on their reliability, recency, and diversity. Results show that the most cited principle in the countries is Transparency. The principles that cover most countries (i.e., global principles) are Privacy, Transparency, Fairness, Security, Safety, Responsibility, Accountability, Explainability, Well-being, Human Rights, and Inclusiveness. Results from the implementation analysis show that United States and United Kingdom are more favorable for investment, business, and research in AI with a good growth stability between 2020 and 2021. They also show the fast growing countries in terms of AI readiness and the current implementation tools (e.g., laws, standards, checklists, training, softwares). The principleimplementation gaps include lack of effective implementation tools and lack of practical training courses on AI. Recommendations for gap mitigation include inclusiveness and

The authors are with Polytechnique Montréal, Montréal, QC H3C 3A7, Canada. E-mail: {lionel.tidjon, foutse.khomh}@polymtl.ca

diversity of team roles, education and awareness on ethical values, and use of EthicsOps. EthicsOps is the continuous planning, execution, and monitoring of ethics principles in a project lifecycle (including AI project). It ensures that team roles are diverse (i.e., no race and gender discrimination), business requirements are pro-ethically aligned (i.e., meet ethics laws and standards, ethics governance guidelines), ethical controls are enforced (e.g., checklists), and that ethics by design practices are applied [8] during the project lifecycle. Governments, national, and international organisations can leverage these results to guide their AI initiatives and conduct ethical AI implementations adapted to their context.

The rest of this paper is structured as follows. Section 2 reviews the relevant literature. Section 3 describes the methodology of our study. In Section 4, we evaluate recent ethical AI principles tailored to specific countries and those that are more generic. Section 5 presents an analysis of current implementations of AI principles. In Section 6, we provide recommendations to mitigate the gaps between AI principles and their implementation. Section 7 discusses threats that could affect the validity of the reported results. Section 8 concludes the paper and outlines some perspectives.

2 RELATED WORK

Floridi *et al.* [4] proposed five ethical principles to lead the development and adoption of AI for the benefit of society and presented 20 concrete recommendations to assess, develop, incentivise, and support good AI. However, the proposed recommendations are high-level and mainly focus on AI ethics governance. This work does not evaluate the application of AI ethics in practice (e.g., country AI readiness analysis, principle-implementation gap analysis).

Jobin *et al.* [5] performed an in-depth analysis of AI guidelines and observed that they are structured around five global principles (transparency, justice and fairness, non-maleficence, responsibility, and privacy). They also identified two important actions to be undertaken by the global community: (1) translating principles into practice and (2) seeking harmonization between AI ethics codes (soft law) and legislation (hard law). However, they have not analyzed ethical AI implementations.

Fjeld *et al.* [7] proposed a visualization of AI principles classified by themes (e.g., International Human Rights, Privacy) and sectors (multi-stakeholder, inter-governmental, private, government, civil society) to provide a high-level snapshot of the current state of AI ethics governance. Similarly to [4], [5], no analysis of ethical AI implementations and principle-implementation gaps was conducted.

Hagendorff *et al.* [6] evaluated 22 guidelines and found that diversity is lacking in the AI community, political abuse of AI systems (e.g., fake news, election fraud, automated propaganda) is also missed by the guidelines, and artificial general intelligence is not discussed. Our work extends [6] with local and global analysis of AI principles (countries, continents), country AI readiness analysis, principle-implementation gap analysis and mitigation.

3 STUDY METHODOLOGY

The goal of this study is to understand the landscape of AI principles around the world, through a contextual and holistic analysis of the characteristics of the different ethical AI principles proposed by countries from different continents, in order to identify potential principleimplementation gaps, their root causes, and formulate recommendations for gap mitigation. The perspective of this study is that of governments, national, and international organisations, which can leverage our findings to guide their AI strategy in developing and implementing AI principles adapted to their context. The context of this study is a set of 100 documents and websites containing 100 AI principles from 29 countries in the 6 continents (e.g., fairness, transparency) and 121 implementations of these principles (e.g., guidelines, standards, tools) from 14 countries in the 6 continents, gathered online from reports of national/international organizations and websites. The data sources are selected based on their reliability, recency, and diversity. We address the following research questions:

RQ₁: What are the characteristics of ethical AI principles across countries and continents? This research question aims to help identify the countries (resp. continents) that released a high number of principles, the principles on which countries (resp. continents) place a greater emphasis (i.e., the most cited principles) and the principles shared across different countries (resp. continents). The common principles found between continents are compared to those from existing international organisations (i.e., OECD, GPAI, G20) to identify missing in the existing principles.

RQ₂: What is the current level of AI readiness in countries as well as the current implementations of the ethical AI principles? This research question seeks to identify the level of preparedness of countries based on scoring indexes [9], [10] to implement the necessary changes required by the adoption and development of AI technologies. We also aim to understand how ethical AI principles are implemented in these countries.

RQ₃: What are the gaps between ethical AI principles and their implementations as well as their root causes? This research question aims to identify any gap that may exist between the stated AI principles and their implementation. We also aim to identify the root causes of these gaps, in order to formulate recommendations for their mitigation.

 \mathbf{RQ}_4 : What are potential solutions for principleimplementation gap mitigation? This research question aims to propose some recommendations, to mitigate gaps between ethical AI principles and their implementation.

3.1 Data collection

Three selection criteria have been defined for data collection: reliability, recency, and diversity. The reliability criteria ensures that the data can be trusted or is provided by a

| | TABLE 1: A record | example of | principles | from all | l continents |
|--|-------------------|------------|------------|----------|--------------|
|--|-------------------|------------|------------|----------|--------------|

| Provider | Title | Location | Number | Principles | References | Date | Туре | Continent |
|---|---|-------------------|--------|---|--|------|-----------------------|---------------|
| The Institute for Ethical AI & Machine Learning | The Responsible Machine Learning Principles | United Kingdom | 8 | Human augmentation, Bias evaluation, Explainability By Justification, Reproducible operations, Displacement strategy, Pratical Accuracy, Trust by privacy, Data risk awareness | https://ethical.institute /principles.htmtl | | Research Institute | Europe |
| Universite de Montreal | Montréal Declaration for Responsible AI | Canada | 10 | well-being, autonomy, intimacy and privacy, solidarity, democracy, equity, inclusion, caution, responsibility and environmental sustainability | https://www.montreal declaration-responsible ai.com/reports-of-montreal -declaration | 2017 | University | North America |
| IEEE | Ethically Aligned Design: A Vision for Prioritizing Human Well-being with Autonomous and Intelligent Systems - General Principles | US | 5 | Human Rights;Prioritizing Well-being;Accountability; Transparency;A/IS Technology Misuse and Awareness of It | https://standards.ieee.org/ content/dam/ieee-standards/ standards/web/documents/ other/ead_v2.pdf | 2018 | Inter national | North America |
| India Government | Ethics and Human Rights | India | 5 | equality, safety & reliability, inclusivity & non-discrimination, transparency, accountability and privacy & security | https://indiaai.gov.in/ research-reports/responsible -ai-part-1-principles-for- responsible-ai | 2021 | Government | Asia |
| AI Forum | Trustworthy AI in Aotearoa | New Zealand | 5 | Fairness and Justice;Reliability, Security and Privacy;Transparency ;Human Oversight and Accountability;Well being | https://data.govt.nz/assets/ data-ethics/algorithm/ Trustworthy-AI-in- Aotearoa-March-2020.pdf | 2020 | Government | Ocenia |
| Research ICT Africa | Recomendations on the inclusion subSaharan Africa in Global AI Ethics | South Africa | 6 | Introduce safeguards to balance AI opportunities and risks. Protect individual and collective privacy rights in crossborder data flows; Define African values for AI and align AI frameworks with such values; Practise fair and socially-responsible AI; Build inclusive partnerships based on community and cocreation; Adopt an adaptive, open minded and humble approach | https://researchictafrica.net/ wp/wp-content/uploads/2020/ 11/RANITP2019-2-AI-Ethics.pdf | 2019 | Research Institute | Africa |
| Chile's Ministry of Science | Cross-cutting AI principles | Chile | 4 | IA with a focus on people's well-being, respect for human rights and security; IA for sustainable development; Inclusive AI; Globalized and evolving AI | https://www.carey.cl/en/ chile-presents-its-first- national-artificial-intelligence -policy/ | 2021 | Government | South America |

trusted sources. More than 350 reports and websites from companies, national, and international organizations have been inspected to collect AI principles and proposed implementation tools. For each candidate report, we assessed the trustworthiness of the institutions behind the report. An institution is considered to be trustworthy if it is officially recognised by an official state (e.g., The White House) and an international organisation (e.g., UNESCO).

The recency criteria ensures that the data source contains recent information about ethical AI principles. From the 350 collected reports and websites, we have selected 100 relevant reports and websites published between 2016 and 2022. To select a report, we first read the document/website and check if it contains a clear reference to ethical principles (e.g., *Our principles, We follow these principles*), or to the implementation of principles (e.g., law and standard references, reference links for ethical AI softwares, training courses on AI ethics). Reports that did not contain ethical AI principles nor implementations were ignored.

The diversity criteria is used to ensure that we collect data from different countries and continents. The final set of reports analyzed in this paper covers countries from Europe, North America, South America, Oceania, Africa, and Asia. The references of these reports can be found in our replication package [11].

3.2 Data extraction

In the hundred reports and websites, we have manually selected 100 ethical AI principles randomly, representing a confidence level of 90% with an error margin of 8.25%. Each principle contains 2 to 15 keywords including Safe, Safety, Transparent, Transparency, Robust, Robustness, Ex-

TABLE 2: Number of AI ethics implementations

| AI Ethics Implementation | Count |
|--------------------------|-------|
| Guidelines | 26 |
| Regulations & Laws | 20 |
| Standards | 8 |
| Governance | 7 |
| Checklists | 6 |
| Algorithmic Assessment | 16 |
| Software | 28 |
| Training | 10 |
| Total | 121 |

plainable, Explainability, Fair, Fairness, Secure, Security, Responsible, Responsibility, Inclusive, and Inclusiveness.

Using Microsoft Excel, we have built regular expressions to extract these string keywords. The following regular expression was used to extract each principle keyword: *Safe* (Safety), *Transparen* (Transparency), *Robust* (Robustness), *Explainab (Explainability), *Fair* (Fairness), *Secur* (Security), *Responsib* (Responsibility), and *Inclusive* (Inclusiveness). For each principle, we extracted and recorded the following information: the name of the report's provider (e.g., US Department of Defense, IEEE), the title of the report, the location where the report is published, the number of principles contained in the report, the contained principles, the reference to get the principles, the date when the principle was released, the type of institution (e.g., government, international, company, university), and continent. An example of recorded information is shown in Table 1. The complete dataset is available in our replication package [11].

We have extracted 521 occurrences of AI principle keywords from 29 countries: Canada, United States (US), United Kingdom (UK), China, Australia, Singapore, United Arab Emirates (UAE), Spain, Switzerland, Chile, Colombia, Sweden, Belgium, Israel, Amsterdam, New Zeland, India, Germany, Hong Kong, South Korea, South Africa, Finland, Japan, France, Netherlands, Norway, Russia, Ireland, and Italy. Table 3 presents the distribution of these principles across the countries. The 521 occurrences are grouped using 33 distinct keywords: Fairness, Transparency, Accountability, Safety, Security, Robustness, Explainability, Interpretability, Well-being, Human Oversight, Human Rights, Sustainability, Equity, Reliability, Privacy, Justice, Autonomy, Human Dignity, Responsibility, Solidarity, Beneficence, Non-maleficence, Non-discrimination, Contestability, Human-centric, Inclusiveness, Trustworthy, Democracy, Governance, Bias, Integrity, Controllability, and Accuracy.

We also extracted 121 AI ethics implementations from the hundred reports and websites (see Table 2), representing a confidence level of 95% with an error margin of 8.91%. In Table 2, the extracted AI ethics implementations are grouped into 8 categories: guidelines, regulations & laws, standards, governance, checklists, assessment, tools, and training. Categories such as guidelines, regulations & laws, standards, and governance are more abstract and they provide orientations and rules to follow for AI ethics implementation. Other categories like checklists, assessment, and tools are more technical and they provide practical solutions for AI ethics implementation. The last category (i.e., training) contains courses about communicating and learning AI ethics.

3.3 Data Processing and Analysis

We processed the extracted data using word frequency analysis. The word frequency analysis is performed by counting the number of AI principle occurrences per country and per continent. The aim being to identify potential characteristics such as the countries (resp. continents) that released a high number of principles, the principles on which specific countries (resp. continents) place a greater emphasis and the principles shared across different countries (resp. continents). Then, findings for the defined research questions are presented through point clouds using RStudio Integrated Development Environment (IDE). Point clouds show small and large circles depending on the number of occurrences of the principle; thus helping to identify the principle emphasis in the country (resp. continent).

4 EVALUATION OF ETHICAL AI PRINCIPLES

In this section, we analyze the processed data and provide answers to our first research question (i.e., RQ1). We organise our analysis in two parts: (1) we perform a *contextual* analysis of ethical AI principles; by identifying and analyzing AI ethical principles defined for specific countries, and (2) We perform a *holistic* analysis by considering ethical AI principles that are generic, i.e., principles stated at a global scale (e.g., continent-level).

4.1 Contextual evaluation

To identify the characteristics of ethical AI principles across countries (RQ1), we have formulated two sub-research questions: What are the countries that released a high number of principles ? and What are the most cited principles in the countries as well as the principles shared across different countries ?

TABLE 3: Number of principles per country

| Country | Total Number of principles |
|----------------------------|----------------------------|
| Canada | 29 |
| United States (US) | 167 |
| United Kingdom (UK) | 53 |
| China | 27 |
| Australia | 12 |
| Singaporo | 12 |
| United Arab Emirator (UAE) | 14 |
| | 4 |
| Spain Cruite autor d | 5 |
| Switzerland | 10 |
| Chile | 4 |
| Colombia | 10 |
| Sweden | 9 |
| Belgium | 12 |
| Israel | 6 |
| Amsterdam | 5 |
| New Zeland | 11 |
| India | 5 |
| Germany | 28 |
| Hong Kong | 3 |
| South Korea | 19 |
| South Africa | 6 |
| Finland | 10 |
| Japan | 25 |
| France | 14 |
| Netherlands | 3 |
| Norway | 7 |
| Russia | 10 |
| Ireland | 9 |
| Italy | 4 |
| Total | 521 |

4.1.1 What are the countries that released a high number of principles ?

The aim of this question is to identify countries which are more involved in releasing national ethical AI principles. In Table 3, United States (US) have the highest number of principles among the 29 countries. The second is United Kingdom (UK). Canada and Germany are respectively in third and fourth position among the 29 countries. Next come China and Japan in fifth and sixth position, respectively, among the 29 countries. In North America, US has released the highest number of principles followed by Canada. Countries like Mexico are not mentioned because they have not yet released any national ethical AI principles. However, Some of the missing countries are part of (or in the process of becoming member of) international organizations such as the Organization for Economic Cooperation and Development (OECD), which will be described later.

In South America, Colombia is the first country to release a national ethical AI principles, followed by Chile. Other South American countries such as Argentina and Brazil do not have national ethical AI principles, but they are in a partnership with OECD. Chile, Costa Rica, Colombia, and Mexico are also OECD members. In Europe, UK is the first country to publish national ethical AI principles, followed by Germany. France appears in the third position followed by Belgium. In Asia, China is the first country with the highest number of principles and Japan occupies the second position. Next come South Korea and Singapore in third and fourth position, respectively. In Oceania, Australia is the first country with the highest number of principles followed by New Zealand. Other countries such as Nauru did not release national principles. In Africa, South Africa is the value • 0 • 25 • 50 • 75 • 100

| | INort | th Amer | rica | | | | | Eu | irope | e | | | | | | | - Oce | ania | | | | | Asia | | | | Sc | outh A | merica | Africa |
|-----|--------------------------|-----------------|--------------|-----|------|--------|------|------|-------|-----|-----|------|-----|------|--------|-------|-------|------------------|-----|--------------|-----|---|------|--------|----------------|--------|--------------|--------|--------|--------|
| | | | | • | | • | • | • | | • | | | • | • | • | • | | • | 1 | • | | | • | • | | | | | • | |
| | u-being | X : | ŏ | • | • | • | • | č | ŏ | • | • | • | • | • | • | • | | ÷ | | | • | • | • | • | • | 0 | 0 | ••• | • | • • |
| | Wein orthy | \sim | ŏ | 0 | 0 | 0 | 0 | 0 | ŏ | 0 | 0 | • | 0 | 0 | 0 | 0 | 0 | 0 | 1 | 0 | 0 | 0 | 0 | 0 | 0 | ŏ | ŏ | | • | • |
| | Trustency | Δ : | õ | • | • | • | • | • | ÷ | • | õ | • | • | • | • | • | | • | • | • | • | • | • | • | • | • | ÷ | | 0 | ••• |
| | Transfability 0 | ğ | • | • | • | • | • | • | • | • | • | • | • | • | • | • | • | • | | • | • | • | • | • | • | • | • | • | • | • |
| | Suste Solidarity • | \bigcirc | \circ | 0 | 0 | 0 | 0 | • | 0 | 0 | • | • | • | • | 0 | • | • • | 0 | • | 0 | • | 0 | 0 | 0 | • | 0 | 0 | • 0 | • | • • |
| | Security 0 | \bigcirc | 0 | • | • | 0 | 0 | • | 0 | 0 | 0 | • | 0 | • | • | • | 0 | • | L | • | 0 | 0 | 0 | 0 | • | 0 | 0 | • | • | 0 |
| | 50.0 | 2: | 0 | • | • | • | • | 0 | 0 | • | 0 | • | 0 | • | • | • | • • | • | : | • | • | • | • | • | • | • | • | • • | • | ••• |
| | Robustria O | \bigcirc | 0 | • | 0 | 0 | • | • | 0 | 0 | 0 | 0 | • | 0 | • | • | • | • | | \mathbf{O} | • | • | 0 | • | • | 0 | 0 | 0 | • | 0 |
| | - sponsibility | 8: | | • | • | • | 0 | • | 0 | • | : | • | • | 0 | • | • | • • | 0 | : | 0 | • | • | • | 0 | • | : | $\dot{\sim}$ | ••• | • | ••• |
| | Restaliancy O | \bigcirc | \mathbf{O} | 0 | • | • | • | • | 0 | 0 | 0 | • | 0 | • | • | • | 0 | 0 | | 0 | • | 0 | 0 | 0 | • | 0 | 0 | 0 | • | 0 |
| | Pri ence | | | • | • | • | • | | | • | • | • | • | : | • | • | | • | i | • | • | • | | : | • | ÷ | • | | • | |
| | maleination | | ě | | • | | : | : | | : | | | : | | : | | | | | ċ | | : | | | : | | | | | |
| ŝ | Non-scrimiustice | | ŏ | • | | • | • | • | • | • | | • | • | • | • | • | | · | İ. | • | · | • | • | • | 0 | • | • | | • | |
| ple | Non-Cr. etability | . | ŏ | • | • | • | • | • | 0 | • | • | • | • | • | • | • | • • | • | | • | • | • | 0 | • | | • | 0 | • • | • | • • |
| nci | Interpredicted Integrity | ŏ | ŏ | • | • | • | • | • | • | • | • | • | • | • | • | • | • | • | | \bigcirc | • | 0 | • | • | • | • | • | | 0 | 0 |
| Pri | eiveness . | ŏ • | Ō | • | 0 | 0 | • | • | • | 0 | • | • | • | • | • | • | ••• | • | • | õ | • | • | 0 | • | • | • | • | • 0 | 0 | • • |
| | Inclus Right . | • | • | • | • | • | • | 0 | 0 | • | • | • | • | • | • | • | • | 0 | 1 | • | • | • | • | • | 0 | • | • | • | • | • |
| | Human oversign | •• | • | • | • | • | • | • | • | • | 0 | ٠ | • | • | • | • | ••• | • | • | • | ٠ | 0 | • | ٠ | • | • | • | ••• | • | ••• |
| | uman Dignic • | • | 0 | 0 | • | 0 | • | • | • | • | ٠ | ٠ | ٠ | • | • | • | 0 | • | | • | 0 | • | • | ٠ | ٠ | • | • | • | • | • |
| | Human-cente . | <u>A</u> : | : | • | 0 | • | • | • | 0 | • | • | • | • | • | • | • | • • | • | • (| \circ | • | 0 | • | • | • | 0 | • | • • | • | • • |
| | Humanernans O | \bigcirc | Q | 0 | • | 0 | 0 | 0 | 0 | 0 | 0 | • | • | • | • | 0 | 0 | 0 | | 0 | 0 | 0 | 0 | • | • | • | 0 | • | • | • |
| | Fairning O | • • | \circ | 0 | • | 0 | • | • | 0 | • | 0 | 0 | • | • | • | 0 | • • | • | : | • | 0 | 0 | 0 | • | • | 0 | • | ••• | • | ••• |
| | cyplaina uity | • | • | • | • | • | • | • | • | • | • | • | • | • | • | • | • | • | | • | • | • | • | • | • | • | • | • | • | • |
| | EM ENACY | $\dot{\bullet}$ | : | • | • | : | • | • | • | • | : | • | : | : | • | • | ••• | • | : | ÷ | • | • | • | • | • | • | : | | • | ••• |
| | Democility | | 0 | • | • | 0 | • | • | • | • | 0 | • | 0 | | • | • | | • | • | | • | | • | • | • | • | 0 | | • | |
| | controlle ability | ă i | ŏ | : | | | : | | | : | | | : | | | | | • | i | | | : | | : | : | | | | • | |
| | Contest Bias | X! | ŏ | | ĕ | | | | | | | | | | | | | | | | | | | | | | | | | |
| | ficence | X i | | | ÷ | | | | | | ŏ | | ŏ | • | | | | | | | | č | • | | | | | | | |
| | Benenionny | | ŏ | • | • | • | ò | • | • | • | | • | ÷ | ÷ | • | • | | • | | • | 0 | • | ÷ | • | • | • | • | | • | |
| | Aurouracy | \frown | ŏ | • | • | 0 | ŏ | • | 0 | • | 0 | • | 0 | • | 0 | 0 | 0 | 0 | 1 | • | ŏ | 0 | 0 | 0 | 0 | 0 | 0 | | • | • |
| | Auntability C | у Г | -¥ | Ŀ | þ | - c | Ť. | Ē | ž | þ | | -s | ž | a- | - b | 7 | a t | - - - | • | a | | ш | - | a a | - ⁶ | a a | Ť | | e | |
| | Accon. be | ä | 5 | pai | lan | ede | giur | dar | nan | lan | anc | and | rwa | ISSI | lan | ltal | trali | llan | | ihi | por | N | srae | Indi | ov | ore | apa | iqu | Chil | fric |
| | Cai | | | 0) | itze | ŠŇ | Bel | stei | 3err | Ē | Ľ. | Jerl | No | Ŗ | Ire | | Aus | Zeć | | 0 | nga | | | | ng l | ξ | ÷ | 00 | - | th⊳ |
| | | | | | Swi | | | Am | Ċ | | | Vetl | | | | | | ew | | | ŝ | | | | Ч | sour | | 0 | | Sou |
| | | | | | | | | | | | | 2 | | | 0 | 20110 | try | z | | | | | | | | 0) | | | | 0) |
| | | | | | | | | | | | | | | | , c | Jour | u y | | | | | | | | | | | | | |

Fig. 1: Number of occurrences per principle per country

first to have national principles via the Research ICT Africa (RIA)¹. Other African's countries do not yet have national ethical AI principles, but some are part of (or in the process of becoming member of) OECD.

4.1.2 What are the most cited principles in the countries as well as the principles shared across different countries?

This research question aims to identify the principles on which countries place greater emphasis (i.e., the most cited principles) and the principles shared across different countries. Fig. 1 shows the number of principle occurrences per country from the set of principles extracted from the selected reports. A high (resp. low) circle diameter means a high (resp. low) number of principle occurrences in the country.

In North America, the Transparency principle has the largest frequency in United States and Canada. Transparency and Privacy principles are both cited in United States and Canada. In United States, we have identified 5 top frequent principles as follows: Transparent, Fairness, Accountability, Privacy, and Security. In Canada, the most frequent principles are Transparency, Responsibility, Privacy, Sustainability, Autonomy, and Well-being.

In Europe, the Transparency principle is the most frequent principle across 13 countries: UK, Germany, France, Belgium, Switzerland, Finland, Sweden, Norway, Spain, Amsterdam, Russia, Ireland, and Italy. It is described in Fig. 1 by gray circle appearing in these countries. The second most frequently occurring principle is Fairness. This principle covers 9 countries: UK, Belgium, Sweden, Germany, France, Spain, Amsterdam, Finland, and Italy. Security, Responsibility and Accountability principles are found in 8 countries (see Fig. 1). The Security principle appears in Canada, US, UK, Spain, Switzerland, Sweden, Belgium, Germany, Finland, and Ireland. The Responsibility principle is found in Germany, UK, Switzerland, Sweden, France, Finland, Russia, and Netherlands. The Accountability principle appears in UK, France, Belgium, Sweden, Germany, Norway, Ireland, and Italy. Explainability and Safety principles appear in 7 countries including UK, Germany, Sweden, and France. The Privacy principle appears in 6 countries including UK, France, Norway, and Germany.

| TADI | T 4 | NT 1 | <i>c</i> | • | | | |
|------|------|------|----------|-----------|-------|-----|------------|
| TABL | E 4: | Num | per of | princ | ples | per | confinent |
| | | | | p i i i c | 10100 | P | contentori |

| Continent | Total Number of principles |
|---------------|----------------------------|
| Europe | 210 |
| Asia | 103 |
| North America | 201 |
| South America | 14 |
| Africa | 6 |
| Oceania | 23 |

In Asia, the Transparency principle is the most frequent principle across 8 countries: Japan, Singapore, South Korea, Israel, India, China, United Arab Emirates (UAE), and Hong Kong (see Fig. 1). The second most frequent principle is Accountability. This principle is found in 7 countries: Japan, Singapore, Israel, South Korea, India, UAE, and Hong Kong. Safety, Security, and Privacy principles appeared in 6 countries. The Safety principle appears in South Korea, Singapore, Japan, Israel, India, and UAE; Security and Privacy principles appear in Japan, China, Israel, South Korea, India, and UAE. Other principles like Fairness, Explainability, and Responsibility are ranked in fourth position and found in 4 countries. The Fairness principle appears in China, Singapore, Israel, and UAE; and the Explainability principle appears in South Korea, Israel, UAE, and Singapore. The Responsibility principle is found in China, Japan, South Korea, and Israel.

Summary 1

- In the studied samples, United States, United Kingdom, and Canada released the highest number of principles between 2016 and 2021.
- Most countries from North America, Europe, Asia, Oceania, and South America placed a greater emphasis on the Transparency principle, identified by a high number of occurrences.
- In North America, Transparency and confidentiality are the principles most frequently cited by countries.
- Transparency, Fairness, and Security are the principles most frequently stated by countries in Europe.
- In Asia, Transparency and Accountability are the most frequently stated principles (by countries).
- Transparency, Fairness, Accountability, Security, Privacy, Reliability, and Well-being are the most frequently cited principles, by the countries in Oceania.
- In South America, Colombia and Chile share principles such as Security (resp. Safety), Human Rights (resp. Human-centric), and Non-discrimination (resp. Inclusiveness).
- In Africa, South Africa is focused on Fairness, Inclusiveness, Safety, Privacy, Responsibility, and Human Rights principles.

Oceania, Australia and New Zealand have 7 principles in common: Transparency, Fairness, Accountability, Security, Privacy, Reliability, and Well-being. In addition, Australia has Explainability, Contestability, and Human-centric principles whereas New Zealand has Human Oversight and

4.2 Holistic evaluation

ples.

In this section, ethical AI principles are analyzed on a global scale to identify the principles on which continents place a greater emphasis (i.e. the most cited principles) and the principles shared across different continents. The identified shared principles are compared with common principles provided by international organizations such as OECD, GPAI, and G20 to which different countries are affiliated. To structure our analysis, we have defined two sub-research questions: *What are the most cited principles in the continents as well as the principles shared across different continents?* and *How do continent-specific principles differ from those provided by international organizations (i.e., OCECD, GPAI or G20)?*.

4.2.1 What are the most cited principles in the continents as well as the principles shared across different continents ?

This research question aims to identify the principles on which continents place greater emphasis (i.e., the most cited principles) and the principles shared across different continents. Fig. 2 shows that Europe and North America have the highest number of ethical AI principles. Human Rights and Privacy principles are shared by all continents. Transparency, Fairness, Security, and Safety principles have the highest number of occurrences and they appear in 5 continents: Europe, North America, Asia, Oceania, South America, and Africa. Responsibility, Well-being, and Inclusiveness principles are also found in 5 continents: Europe, North America, Asia, South America, Oceania, and Africa. Accountability and Explainability principles appear in 4 continents with a high number of occurrences: Europe, North America, Asia, and Oceania.

Global principles. We identified 11 global principles that cover most continents with a high number of occurrences: Transparency, Privacy, Fairness, Security, Safety, Responsibility, Accountability, Explainability, Well-being, Human Rights, and Inclusiveness.

4.2.2 How do continent-specific principles differ from those provided by international organizations such as OCECD, GPAI or G20 ?

In this research question, we aim to examine the alignment between the global principles developed on different continents and those proposed by international organizations (i.e., OECD, GPAI, G20).

4.2.2.1 GPAI: GPAI² is a multi-stakeholder association consisting of scientists, industries, civil society, governments, international organisations and academia aiming to bridge the gap between theory and practice on AI. They

2. https://gpai.ai/



Fig. 2: Number of occurrences per principle per continent

released 5 principles: (1) Inclusive growth, sustainable development and well-being; (2) Human-centred values and fairness; (3) Transparency and explainability; (4) Robustness, security, and safety; and (5) Accountability. The 11 global principles identified in the previous question are similar to these principles. Well-being, Human Rights, Inclusiveness, and Fairness principles cover principles (1) and (2), except the Sustainability principle. Security, Safety, Transparency, Explainability, and Accountability principles cover principles (3), (4) and (5), except the Robustness principle. The Responsibility principle encompasses Sustainability, Robustness, reliability, impartiality). GPAI principles do not contain the Responsibility principle which is important (it appears in 5 continents with a high number of occurrences).

4.2.2.2 OECD: OECD³ is an international organization for economic studies with country members from all continents. OECD promotes 5 AI Principles: (1) Inclusive growth, sustainable development and well-being; (2) Human-centered values and fairness; (3) Transparency and explainability; (4) Robustness, security and safety; and (5) Accountability. From the identified global principles, (1) is covered by Inclusiveness, Well-being, and Responsibility principles. Principle (2) is covered by Human Rights and Fairness. (3) and (4) are covered by Transparency, Security, Safety, Responsibility, and Explainability principles. Principle (5) is also found in the global principles. OECD AI principles do not contain Privacy and Responsibility principles; yet they appeared in 5 continents with a high number of occurrences.

4.2.2.3 G20: G20⁴ is an intergovernmental forum with 19 country members and the European Union, that addresses global economy issues (e.g., climate change). G20 AI principles are drawn from the OECD AI principles. Thus, G20 AI principles do not also contain Privacy and Responsibility principles.

Summary 2

- According to our analysis, North America and Europe released the highest number of principles between 2016 and 2021.
- The 11 global principles that cover most continents with a high number of occurrences are: Privacy, Transparency, Fairness, Security, Safety, Responsibility, Accountability, Explainability, Wellbeing, Human Rights, and Inclusiveness.
- GPAI ethical AI principles do not include the Responsibility principle.
- OECD and G20 ethical AI principles do not include Privacy and Responsibility principles.

5 EVALUATION OF ETHICAL PRINCIPLE IMPLEMEN-TATIONS

In order to identify gaps between ethical AI principles and their implementations, it is important to review and understand the current implementations in the countries. In [9], [10], Oxford Insights and International Development Research Centre (IRDC) provide information about the AI readiness of countries in terms of governance, infrastructure, and technology for AI implementation support. Based on the information contained in [9], [10], we perform a country AI readiness analysis between 2020 and 2021, to identify the current progression state of AI implementations in the different countries. Next, we perform an in-depth analysis of recent ethical AI implementations documents, such as guidelines, code of conduct (i.e., regulation and laws, standards), governance rules, ethical AI assessment tools (i.e., checklists, algorithmic assessments), implementation tools, and awareness activities (i.e., online training, campaigns). In the following, we report about these analysis; presenting the level of preparedness of different countries (i.e., their AI readiness) and the existing ethical AI implementations (RQ2). We also report about gaps identified between published ethical AI principles and their implementation (RQ3).

5.1 What is the current level of AI readiness in countries?

In order to identify the country AI readiness (RQ2), scoring indexes [9], [10] evaluates the current level of AI preparedness following different levels: (1) government, (2) technol-

4. https://g20.org/

| TABLE 5: country | AI Readiness score | (2021) |
|------------------|--------------------|--------|
|------------------|--------------------|--------|

| Country | Government | Technology | Data& Infra. |
|-------------------|------------|------------|--------------|
| US | 88.46 | 83.31 | 92.71 |
| Singapore | 94.88 | 66.69 | 85.8 |
| UK | 85.69 | 67.26 | 90.81 |
| Finland | 88.45 | 63.85 | 85.4 |
| Netherlands | 80.42 | 66.17 | 88.92 |
| Sweden | 80.76 | 67.37 | 86.36 |
| Canada | 84.36 | 63.75 | 85.08 |
| Germany | 78.04 | 67.68 | 86.07 |
| Denmark | 83.5 | 63.24 | 84.14 |
| Republic of Korea | 85.27 | 58.49 | 85.89 |
| France | 82.1 | 60.61 | 86.53 |
| Japan | 81.9 | 59.31 | 87.32 |
| Norway | 84.24 | 59.25 | 84.91 |
| Australia | 83.79 | 57.07 | 85.37 |
| China | 83.79 | 61.33 | 78.15 |
| Luxembourg | 82.67 | 50.66 | 86.8 |
| Ireland | 74.7 | 61.11 | 82.59 |
| Taiwan | 77.59 | 59.42 | 78.92 |
| UAE | 79.41 | 53.33 | 82.05 |
| Israel | 64.64 | 65.87 | 79.52 |

| Country | 2020 | 2021 | Growth |
|-------------------|--------|-------|--------|
| US | 85.479 | 88.16 | 2.681 |
| Singapore | 78.704 | 82.46 | 3.756 |
| UK | 81.124 | 81.25 | 0.126 |
| Finland | 79.238 | 79.23 | -0.008 |
| Netherlands | 75.297 | 78.51 | 3.213 |
| Sweden | 78.772 | 78.16 | -0.612 |
| Canada | 73.158 | 77.73 | 4.572 |
| Germany | 78.974 | 77.26 | -1.714 |
| Denmark | 75.618 | 76.96 | 1.342 |
| Republic of Korea | 77.695 | 76.55 | -1.145 |
| France | 73.767 | 76.41 | 2.643 |
| Japan | 73.303 | 76.18 | 2.877 |
| Norway | 74.43 | 76.14 | 1.71 |
| Australia | 73.577 | 75.41 | 1.833 |
| China | 69.08 | 74.42 | 5.34 |
| Luxembourg | 72.616 | 73.37 | 0.754 |
| UAE | 72.395 | 71.6 | -0.795 |
| Israel | 68.825 | 70.01 | 1.185 |

ogy factor, and (3) data and infrastructure. At the government level, the score gives insights about the willingness of the government of the country to adopt AI, and its ability to adapt and innovate. In the technology sector, the score gauges the level of supply of AI tools in the country. At the data and infrastructure level, the score provides insights about the implementation of AI tools and assessing whether they are trained on high quality and representative data, and whether the government of the country has the appropriate infrastructure for delivering AI solutions and fostering its adoption by citizens [10].

Table 5 shows the AI readiness score for 20 countries in 2021: US, Singapore, UK, Finland, Netherlands, Sweden, Canada, Germany, Denmark, Republic of Korea, France, Japan, Norway, Australia, China, Luxembourg, Ireland, Taiwan, UAE, and Israel. At the government level, Singapore comes in first position (94.88) followed by US (88.46). Finland is ranked in the third position (88.45) followed by UK (85.69). The Republic of Korea and Canada are respectively in fifth (85.27) and sixth (84.36) position. In the technology sector, US leads with a score of 83.31, followed by Germany with a score of 67.68. Next is Sweden, ranked in third position (67.37), followed by UK (67.26). Singapore and Netherlands are respectively in fifth (66.69) and sixth position (66.17). At the data and infrastructure level, US is ranked first (92.71) followed by UK (90.81). Netherlands is ranked in third position (88.92) followed by Japan (87.32). Next is Luxembourg and France appearing respectively in fifth (86.8) and sixth position (86.53). From these results, US, UK, and Germany are more favorable for AI development, investment, business, and research since they are well prepared in terms of AI technology, data and infrastructure, and AI governance. Other countries such as Finland, Sweden, Netherlands, Japan, and Singapore are also propice for investment, business, and research in AI.

Table 6 shows the yearly AI readiness score of 18 countries: US, Singapore, UK, Finland, Netherlands, Sweden, Canada, Germany, Denmark, Republic of Korea, France, Japan, Norway, Australia, China, Luxembourg, UAE, and Israel. Overall, US and UK respectively maintained the

first position (+2.681 increase) and second position (+0.126 increase) from 2020 to 2021. Singapore moved from the sixth position in 2020 to the second position in 2021 with a +3.756 increase. Finland moved from the third position in 2020 to the fourth position in 2021 with a -0.008 decrease. Similar to Finland, Germany's readiness score experienced a significant decrease of -1.714 from 2020 to 2021. Sweden's AI readiness score also decreased from 78.772 in 2020 to 78.16 in 2021 (-0.612 decrease). Netherlands have a +3.213 increase from 2020 to 2021. Canada have experienced a high growth from 2020 to 2021 with a +4.572 increase. China have the highest growth with a +5.34 increase from 2020 to 2021. Other countries like France, Japan, Norway, Australia, Israel, and Luxembourg also experienced a significant growth (as shown in Table 6). From these results, China, Canada, Singapore, and Netherlands have a high potential of increase in the next years in terms of AI governance, technology, data and infrastructure. This also makes them favorable for investment, business, and research in AI.

Summary 3

- At the government level, Singapore was the leader followed by US in 2021. In the technology sector, United States was the leader followed by Germany in 2021. At the data and infrastructure level, US was the leader followed by UK in 2021. Then, US, UK, and Germany are more favorable for investment, business, and research in AI.
- In terms of AI readiness, United States maintained the first position followed by United Kingdom from 2020 to 2021. China and Canada experienced the highest growth (resp. +5.34 and +4.572 increase) from 2020 to 2021. The AI readiness of Germany and Republic of Korea decreased significantly (resp. -1.714 and -1.145) from 2020 to 2021. Then, China and Canada have a high potential of increase in the next years, also making them favorable for investment, business, and research in AI.

5.2 What are the current ethical AI principle implementations?

Several ethical AI guidelines, regulation, and tools have been proposed for designing, and implementing ethical AI systems. The ethical principle implementation materials found through our review (described in Sections 5.2.1-5.2.6) are divided in 6 categories: guidelines, code of conduct (e.g., standards), governance tools, assessment tools (e.g., checklists), AI ethics softwares, and awareness materials (e.g., training courses). Guidelines, code of conduct, and governance tools provide high-level implementations of ethical AI principles while awareness materials, assessment tools, and softwares are more technical. Here after, the existing ethical AI implementation tools are presented (RQ2).

5.2.1 AI Ethics Guidelines

Table 7 shows 26 ethical AI guidelines ranging from 2017 to 2022 and classified by name, author, organization, year, and location. T. Hagendorff [6] analyzed 22 guidelines and found omissions in these guidelines. They were not included in the 26 guidelines since the paper [6] itself provides a summary of those guidelines. T. Hagendorff found that artificial general intelligence was not discussed due to the fact that the guidelines were mostly written by research groups with computer science background. He also observed that diversity among people and technology is lacking in the AI community. He also found that very few guidelines was provided to avoid AI systems being used for political abuses (e.g., fake news, election fraud, automated propaganda).

J. Morley *et al.* [12] built a typology to apply ethics at each stage of the Machine Learning (ML) development pipeline by identifying the tools and methods available, and developers or companies researching and producing them. B.C. Goehring *et al.* [13] proposes to educate students and professionals by adding AI ethics in business and law schools, computer science programs, technical organizations; setting guidelines and standards at national, supranational, and global levels; and creating an unified approach to ensure a robust, complete, and holistic assessment of present and future AI ethics implications.

A. Kaminski *et al.* [14] proposes to apply the Values, Criteria, Indicators, and Observables (VCIO) model to AI ethics. The ethical values defined include Transparency. For transparency, the criteria as well as indicators include the disclosure of the origin of the datasets (e.g., an indicator can be whether data is documented) and the disclosure of properties of algorithm/model used (e.g., an indicator can be whether the model has been tested/used before). Then, observables can be a comprehensible logging of all training and operating data, version control of datasets, and wide usage and testing of the model both in theory and practice.

D. Leslie [15] describes constraints that an AI project must follow and explain how to practically build an ethical platform. An AI project must be ethically permissible, fair and non-discriminatory, worthy of public trust, and justifiable. Their proposed ethical platform has three building blocks: (1) Support, Underwrite, and Motivate (SUM) ethical values which are based on the notions of Respect, Connect, Care, and Protect; (2) Fairness, Accountability, Sustainability, and Transparency (FAST) actionable principles which provides the moral and practical tools to deliver safe and reliable AI solutions; and a (3) process-based governance (PBG) framework to operationalize the SUM Values and the FAST Track Principles across the entire AI project delivery workflow.

A.H. Nelson *et al.* [16] recommands to normalize, organize, and operationalize racial equity throughout data integration. They suggest an ongoing process at each stage of the data life cycle-planning, i.e., data collection (e.g., shared data, inclusive data), data access (open data, restricted data, available data), use of algorithms and statistical tools (i.e., following Responsibility, Explainability, Accuracy, and Auditability, Fairness principles), data analysis (e.g., using participatory research to bring multiple perspectives to the interpretation of the data), reporting and dissemination (e.g., providing clear documentation of the data analysis process with analytic files for reproducibility).

The AI High Level Expert Group (HLEG) [17] proposed technical and non-technical methods to implement AI ethics. Technical methods include architectures for Trustworthy AI, ethics and rules of law by design (e.g., transparency-by-design, fairness-by-design), explanation methods (e.g., IBM XAI), testing and validation (e.g., red teaming, bug bounty programs). Non technical-methods include regulation, codes of conduct, standardization, certification, accountability via governance frameworks, diversity and inclusive design teams, education and awareness to foster an ethical mind-set, stakeholder participation and social dialogue.

The Foundation for Best Practices in Machine Learning (FBPML) [18] provides technical and organisational best practices. These organisational best practices include a clear definition and diversity of team roles, determination and definition of an appropriate, feasibility, and solvability of the business problem, determination of the most desirable and feasible model to achieve the desired product outcomes, management and monitoring of product pipeline during the product management lifecycle (i.e., Design, Exploration, Experimentation, Development, Implementation, Operationalisation). Technical best practices include the use of fairness and non-discrimination in products and models, data quality control, representativeness of the product model and data, performance and robustness of model outcomes, monitoring and maintenance of products and models, and explainability of model functions and outputs.

The World Economic Forum (WEF) [19] proposes data governance guidelines including the use of procurement processes with focus on outlining problems and opportunities, the definition of the public benefit of using AI while assessing risks, inclusion of procurement's within a strategy for AI adoption across government, highlighting the technical and ethical limitations of using the data to avoid bias issues (e.g., data quality such as data completeness, representativenes and accuracy), working with a diverse and multidisciplinary team, focusing throughout the procurement process on mechanisms of accountability and transparency norms. Other ethical AI guidelines were also proposed by international organizations such as IEEE [20], OECD [21], International Association of Privacy Professionals (IAPP) [22], Atomium-European Institute for Science, Media and Democracy (EISMD) [23], Hub for Next

| Name | Author | Organization | Year | Location |
|--|-------------------------------|---------------------------------------|------|-------------|
| The Ethics of AI Ethics - An Evaluation of Guidelines | T. Hagendorff | University of Tübingen | 2020 | Germany |
| Understanding artificial intelligence ethics and safety | D. Leslie | The Alan Turing Institute | 2019 | US |
| The global landscape of AI ethics guidelines | A. Jobin et al. | ETH Zurich | 2019 | Switzerland |
| Ethics Guidelines for Trustworthy AI | AI HLEG | European Commission | 2019 | EU |
| From What to How: An Initial Review of Publicly Available AI Ethics Tools, Methods and Research to Translate Principles into Practices | J. Morley et al. | Oxford Internet Institute | 2020 | UK |
| From Principles to Practice: An interdisciplinary framework to operationalise AI ethics | S. Hallensleben C. Hustedt | AIEI Group | 2020 | Germany |
| Building Ethics into AI: Lessons Learned from Pioneers in the Trenches | K. Baxter | Salesforce Research | 2019 | US |
| Technical and Organizational Best Practices | | FBPML | 2021 | Netherlands |
| Advancing AI ethics beyond compliance From principles to practice | B. C. Goehring et al. | IBM | 2020 | US |
| Guidelines for AI Procurement | | World Economic Forum | 2019 | Switzerland |
| Guidance on AI and data protection | | Information Commissioner Office | 2020 | UK |
| Guidance on the AI auditing framework | | Information Commissioner Office | 2020 | UK |
| Building Ethics into Privacy Frameworks for Big Data and AI | | IAPP | 2018 | US |
| Ethical by Design: Ethics Best Practices for Natural Language Processing | J.L. Leidner | Thomson Reuters | 2017 | UK |
| Ethical OS Toolkit | | IFTF & Omidyar Network | 2018 | US |
| Data preprocessing techniques for classification without discrimination | F. Kamiran et al. | Eindhoven University of Technology | 2012 | Netherlands |
| Responsible Design Assistant | | Responsible AI institute | 2020 | US |
| AI RFX Procurement Framework | | Institute for Ethical AI & ML | 2018 | UK |
| Ethically Aligned Design v1 and v2 | | IEEE | 2017 | US |
| A Toolkit for Centering Racial Equity Throughout Data Integration | A. H. Nelson et al. | AISP | 2020 | US |
| Design Ethically Toolkit | Kat Zhou | University of Cambridge | 2016 | UK |
| Responsible AI – Key Themes, Concerns & Recommendations for European Research and Innovation Summary of Consultation with Multidisciplinary Experts | S. Taylor et al. | HUB4NGI | 2018 | EU |
| Ethical Toolkit for Engineering/Design Practice | S. Vallor | Markkula Center | 2018 | US |
| Tools and Ethics for Applied Behavioural Insights: The BASIC Toolkit | | OECD | 2019 | France |
| AI4People—An Ethical Framework for a Good AI Society: Opportunities, Risks, Principles, and Recommendations | L. Floridi | Atomium-EISMD | 2018 | EU |
| Ethics Canvas | | ADAPT Centre & Trinity College | 2017 | Ireland |

Generation Internet (HUB4NGI) Consortium [24]; national organizations such as the UK's Information Commissioner Office [25], [26]; companies such as Thomson Reuters [27] and Markkula Center [28]; and universities such as the University of Cambridge [29], Trinity College [30], and ETH Zurich [5].

5.2.2 AI Ethics Code of Conduct

The code of conduct describes regulatory norms, laws, and standards defined by governments and international organizations.

5.2.2.1 **Regulations & Laws**: Table 8 shows 20 ethical AI regulations and laws ranging from 1974 to 2018 and classified by name, country, and year of publication. In Canada, AI ethics is supported by the protection of personal information laws such as the Privacy Act (1985) [31], the Personal Information Protection and Electronic Documents Act (2000) [32], and the Financial Consumer Agency of Canada Act (2001) [33]. In US, healthcare data and personal information is protected by the Health Insurance Portability and Accountability Act of 1996 (HIPAA) [34]. The protection of personal information laws include Privacy Act (1974) [35], Privacy Protection Act (1980) [36], and California Consumer Privacy Act (2018) [37]. The EU-U.S. and Swiss-U.S. Privacy Shield (2016) [38] law protects personal data transfer between EU and US. US Finan-

| Name | Country | Year |
|--|-----------|------|
| Privacy Act | Canada | 1985 |
| The Personal Information Protection | Canada | 2000 |
| and Electronic Documents Act (PIPEDA) | Curiada | 2000 |
| Financial Consumer Agency of Canada Act | Canada | 2001 |
| Health Insurance Portability and | US | 1006 |
| Accountability Act (HIPAA) | 03 | 1990 |
| Privacy Act | US | 1974 |
| Privacy Protection Act | US | 1980 |
| California Consumer Privacy Act (CCPA) | US | 2018 |
| EU-U.S. and Swiss-U.S. Privacy Shield | US | 2016 |
| Gramm–Leach–Bliley Act | US | 1999 |
| Fair Credit Reporting Act | US | 2018 |
| Data Protection Act | UK | 2018 |
| General Data Protection Regulation (GDPR) | EU | 2016 |
| Privacy Act | Australia | 1988 |
| Personal Information Protection Law (PIPL) | China | 2021 |
| Cyber Security Law (CSL) | China | 2016 |
| Personal Data Protection Act | Singapore | 2012 |
| Protection from Online Falsehoods and | Singaporo | 2010 |
| Manipulation Act (POFMA) | Jingapore | 2019 |
| Protection of Personal Information Act | South | 2013 |
| rotection of reisonal information Act | Africa | 2015 |
| Data Protection Law | Dubai | 2021 |
| Bundesdatenschutzgesetz (BDSG) | Germany | 2009 |

cial institutions have to comply Fair Credit Reporting Act (2018) [39] and Gramm–Leach–Bliley Act (1999) [40] for the collection, dissemination, and use of consumer information (e.g., credit reports). Other laws were proposed by UK (i.e., Data Protection Act [41]), EU (i.e., General Data Protection Regulation [42]), Australia (i.e., Privacy Act [43]), China (i.e., Personal Information Protection Law [44], Cyber Security Law [45]), Singapore (i.e., Personal Data Protection Act [46], Protection from Online Falsehoods and Manipulation Act [47]), South Africa (i.e., Protection of Personal Information Act [48]), Dubai (i.e., Data Protection Law [49]), and Germany (i.e., Bundesdatenschutzgesetz [50]).

5.2.2. **Standards**: Table 9 presents 8 ethical AI standards ranging from 2017 to now and classified by name, organization, and year of publication. The Open Community for Ethics of Autonomous and Intelligent Systems (OCEANIS) [51] is a global AI standards repository that captures AI and Autonomous and Intelligent Systems standards from different organizations such as the British Standards Institution (e.g., BS 8611:2016 [52]), CIO Strategy Council (e.g., CAN/CIOSC 101:2019 [53]), IEEE Standard Association (e.g., IEEE 1232.3-2014 [54]), and ISO/IEC Joint Technical Committee (e.g., ISO/IEC TR 24028:2020 [55]). ISO/IEC JTC 1/SC 42 [56] committees are working on norms for AI including data (ISO/IEC JTC 1/SC 42/WG 2), fairness (ISO/IEC JTC 1/SC 42/WG 3), information security and data privacy (ISO/IEC JTC 1/SC 42/AHG 4).

The IEEE Global Initiative on Ethics of Autonomous and Intelligent Systems also provides several standards including the model process for addressing ethical concerns during system design (IEEE 7000-2021 [57]), transparency of autonomous systems (IEEE 7001-2021 [58]), and ontological standard for ethically driven robotics and automation systems (IEEE 7007-2021 [59]). In 2021, NIST released a standard draft [60] for AI and user trust to help build trustworthy systems and understand user trust in AI, to minimize the risks and achieve benefits. The Association of Computer Machinery (ACM) Code of Ethics and Professional Conduct [61] provides general principles, professional responsibilities, and professional leadership principles. These principles provide a basis/fundamental for ethical decisionmaking applied to a computing professional's conduct. Other national standards were defined by the Australian Computer Society (i.e., Code of Ethics and Professional Conduct) and the UK Committee on Standards in Public Life (i.e., Artificial Intelligence and Public Standards).

5.2.3 AI Ethics Governance

Ethics in corporate governance ensures trust and transparent in the exercise of power by making it credible not only to the shareholders, but also to stakeholders globally (e.g., employees, clients, suppliers) [62]. Table 11 presents 7 the ethical AI governance documents ranging from 2018 to 2021 and classified by name, author, organization, and year of publication. In 2019, the Berkman Klein Center for Internet and Society [63] highlighted the following five key areas in which action is needed for ethics and governance of AI: the public dialogue about information quality of AI ethics, support of the development of inclusive AI governance frameworks, engagement of companies in developing socially beneficial paths for AI technology, and development of new approaches to teaching students and the public at large about the social implications of AI technology.

The European Parliament [64] also reviewed governance frameworks for algorithmic accountability and transparency such as principles vs. rules based governance, governance mechanisms (e.g., co-regulation by standards like GDPR, state intervention like funding research on transparent and accountable algorithms), and existing governance proposals such as algorithmic impact assessment-based solution for accountability of algorithmic systems used by public authorities. They observed that most organisations rely on principles vs. rules that are more risks-oriented (i.e., minimization of harms/impact arising from the use of the AI technology), using impact assessment tools and they recommended a strong coordination between agencies when regulating algorithms for synchronization. A report on AI governance was also released by the University of Oxford [65]. This report presents key points for an ideal AI governance: values and principles such as security and autonomy, and institutional mechanisms such as the ability to provide security, ensure safe AI, and ensure resilience to concept drift and hijacking.

In 2020, the school of governance of the Technical University of Munich (TUM) [66] analyzed the relative weaknesses of major AI users and developers in terms of AI governance (e.g., heterogeneity in the multitude of principles and ethical AI solutions proposed in by organizations and governance) and identified three elements that are important for a proper embedding of ethics in AI governance: (1) hard and soft frameworks for regulating different AI use cases need to be distinguished by policy makers, (2) a global collaboration on defining minimum standards for AI must be done by stakeholder groups and policy makers, and (3) potential conflicts between different principles of AI ethics and the perceptions/preferences of stakeholders on them must be taken into account by corporate decision and policy makers. For ethical AI and autonomous system governance, the University of the West of England [67] states that all

| Name | Organization | Year |
|--|--|--------------|
| Global AI Standards Repository | OCEANIS | 2019-current |
| Ethical design and use of automated decision systems (CAN/CIOSC 101:2019) | CIO Strategy Council | 2019-present |
| ISO/IEC JTC 1/SC 42 Standards for Artificial Intelligence | ISO | 2017-current |
| Artificial Intelligence and User Trust (NISTIR 8332) | NIST | 2021 |
| Artificial Intelligence and Public Standards | Committee on Standards in Public Life UK | 2020 |
| The IEEE Global Initiative on Ethics of Autonomous and Intelligent Systems | IEEE SA | 2019 |
| Code of Ethics and Professional Conduct | ACM | 2018 |
| Code of Ethics and Professional Conduct | Australian Computer Society | |

TABLE 10: AI ethics algorithmic assessment tools

| Name | Organization | |
|------------------------------------|---------------------------|--|
| AI Ethics Assessment Toolkit | Open Roboethics | |
| AI Ethics Assessment Toolkit | Institute | |
| Algorithmic Impact Assessment | Canada | |
| | Platform for the | |
| Ai impact Assessment | Information Society | |
| Ethics & Algorithms Toolkit | San Francisco City | |
| Algorithmic Impact Assessments: | | |
| A Practical Framework for Public | AI Now | |
| Agency Accountability | | |
| Algorithmic Accountability | ALNOW | |
| Policy Toolkit | AI NOW | |
| Algorithmic Equity Toolkit | ACLU of Washington | |
| Ethical Solf accomment Tool | Centre for Applied | |
| | Data Ethics | |
| AI System Ethics | Smart Dubai | |
| Self-Assessment Tool | Sinart Dubui | |
| Ethical Assessment | Data for Children | |
| | Collaborative with UNICEF | |
| Algorithmic Impact Assessment | Data Society | |
| for the Public Interest | | |
| The Assessment List on | European Commission | |
| Trustworthy Al | HLEG AI | |
| Ethics Self-Assessment Tool | UK Statistics Authority | |
| AI System Ethics | University of Colorado | |
| Self-Assessment Tool | Law School Legal Studies | |
| | Research | |
| Fair, Transparent, and Accountable | | |
| Algorithmic Decision-Making | L. Bruno et al. | |
| Processes | | |
| Self-Reflection Tool | Kesponsible Research | |
| | and Innovation | |

systems must be subjected to an ethical risk assessment following standards (e.g., BS 8611:2016) and then must be redesigned to reduce the impact of any ethical risks (e.g., redesign sub-systems for securely logging system inputs, outputs and decisions).

In 2019, AI4People [23] also proposed a Scalable, Modular, Adapted, Reflexive, and Technologically-savvy (SMART) coordination model using priority groups such as no-regret actions (i.e., new standards and procedures for AI), forms of engagement (e.g., new forum for discussion and consultation), and coordination mechanisms such as de-regulated special zones (i.e., kinds of living environments for the empirical testing and development of AI and robotics), in order to make rational decisions for critical issues.

5.2.4 AI Ethics Assessment

AI ethics assessment tools can be checklists or algorithmic processes that allows one to evaluate whether AI systems are ethically aligned. 5.2.4.1 **Checklists**: Table 12 shows 6 ethical AI checklists ranging from 2007 to now and classified by name, organization, and year of publication. Deon [68] helps to easily build ethics checklist for data science projects. In 2019, the Carnegie Mellon University (CMU)'s Software Engineering Institute [69] proposed checklists with agreement topics (e.g., we will design our AI system with the following ethical principle in mind, we value respect and security) to guide the development of accountable and trustworthy AI systems.

In 2020, Microsoft [70] also proposed an AI fairness checklist to address fairness issues in different stages of the AI project lifecycle: planning meetings (e.g., envision system and scrutinize system vision for potential fairnessrelated harms), reviews (e.g., define and scrutinize system architecture considering potential fairness-related harms), prototyping (e.g., fairness evaluation in system/dataset prototype), building (e.g., fairness evaluation in production datasets/code/system), launching (e.g., revision of systems to mitigate any harms), and evolution (e.g., continuous updates of checklists, continuous monitoring of fairness in systems to identify and correct new biaises).

In 2021, AlgorithmWatch [71] also defined two checklists (i.e., transparency triage and transparency report) to help identify which ethical transparency issues must be addressed/documented before, during, and after the usage of an automated decision-making system w.r.t ethic principles such as justice, fairness, privacy, cybersecurity, autonomy, and transparency.

In 2007, the Center For Research Libraries [72] proposed checklists for the audit and certification of organizational infrastructures, digital object management, technologies, technical infrastructure, and security based on evidence, transparency, adequacy, and measurability criteria. This checklists help to address the trustworthiness of data management systems (including digital repositories), which is important to ensure the safety and security of data processed by AI systems. The Employer Assistance and Resource Network (EARN) [73] on disability inclusion also defined a set of questions for leadership (e.g., compliance with diversity and inclusion policies), human resources personnel (e.g., inclusion of individuals with disabilities), equal employment opportunity managers (e.g., equity and diversity inclusion), and procurement officers (e.g., nondiscrimination/fairness for people with disabilities). These questions on diversity and inclusion are important for AI governance in order to foster ethical AI systems.

5.2.4.2 **Algorithmic assessment**: Table 10 shows 16 ethical AI algorithm assessments found in the literature, classified by name and description. The Open Roboethics

| Name | Author | Organization | Year |
|--|------------------------|--------------------------|---------|
| ML Cards for D/MLOps Governance | I. Hellström | Databaseline | 2021 |
| ALEthics and Covernance "Building a Connected Intelligent and Ethical World" | C Lütao | TUM School of | 2020 |
| Ai Ethics and Governance. Dununig a Connected, intelligent and Ethical World | C. Luige | Governance | |
| Ethics and Covernance of AI | | Berkman Klein Center | 2019 |
| | | for Internet and Society | 2017 |
| A governance framework for algorithmic accountability and transparency | A. Koene et al. | European Parliament | 2019 |
| Machine Ethics: The Design and Governance of Ethical AI | | University of the | 2019 |
| and Autonomous Systems | A. I. Withfield et al. | West of England | nd 2017 |
| On Good AI Governance 14 Priority Actions, a S.M.A.R.T. Model of Governance, | | AI4People | 2019 |
| and a Regulatory Toolbox | 0. i agano | Лін соріс | 2017 |
| AI Governance: A Research Agenda | A. Dafoe | University of Oxford | 2018 |

TABLE 12: AI ethics checklists

| Name | Organization | Year |
|--|------------------------------------|--------------|
| Deon | Driven Data | 2018-current |
| AI Fairness Checklist | Microsoft | 2020 |
| AlgorithmWatch | AlgorithmWatch | 2021 |
| Checklist for Employers: Facilitating the Hiring of People with Disabilities Through | EARN | 2020 |
| the Use of eRecruiting Screening Systems, Including AI | | |
| Designing Ethical AI Experiences Checklist and Agreement | CMU Software Engineering Institute | 2019 |
| Trustworthy Repositories Audit & Cortification: Criteria and Checklist (TRAC) | Center For Research | 2007 |
| indsworting Repositories Addit & Certification. Chieffa and Checklist (TRAC) | Libraries | |

Institute [74] proposed an algorithmic assessment tools consisting in 3 phases: identification of use cases and stakeholders (e.g., use of machine learning algorithm for better prediction of nanny profiles, professional nanny as stakeholder), ethic risk discovery (e.g., using societal values such as transparency, trust, fairness, diversity, human rights), and creation of an implementation road map w.r.t the societal values. The Canadian government also built an algorithmic impact assessment tool [75] to mitigate the impact associated to the use of an automated AI decision system.

The ECP Platform for the Information Society [76] developed steps to conduct an AI impact assessment (e.g., determine the need to perform an AI assessment, describe the AI application and its benefits, checking whether application is reliable/safe/transparent). San Francisco City provides an algorithm toolkit⁵ consisting of two parts: an algorithm risk assessment (e.g., impact, data use risk, accountability risk, third-party methodology risk, historic bias risk, technical bias risk) and an algorithm risk management that matches risk to mitigation, using various severity levels such as very low, low, moderate, high, and extreme.

The AI Now Institute [77] also released an algorithmic impact assessment process consisting of 5 steps: pre-acquisition (e.g., procurement planning, contract negotiation and signing), disclosure requirements (e.g., automated decision system disclosure, potential harms/bias/inaccuracy evaluation), comment period (e.g., public engagement to the process), due process challenge period (e.g., the public challenges the process to identify weaknesses), and impact assessment renewal. Other algorithmic assessment tools were defined by the American Civil Liberties Union (ACLU) of Washington [78], Smart Dubai [79], the Data for Children Collaborative with UNICEF [80], the Data Society [81], the European Commission High-Level Expert Group (HLEG) AI [82], the UK Statistics Authority [83], the University of Colorado Law School Legal Studies Research [84], and the European Commission Responsible Research and Innovation [85].

5.2.5 AI Ethics Software

Table 13 shows 28 ethical AI softwares found in the literature, classified by name and description. Aequitas [86] is a fairness audit software that allows one to audit machine learning (ML) models for discrimination and bias to make equitable decisions. Deon [68] is another tool that allows one to build ethical checklists that an AI application must follow. Fairness, Accountability, and Transparency (FAT) Forensics [87] is a toolbox for the evaluation of FAT principles in predictive systems. AI Fairness 3606 is tool of metrics developed by IBM to check, report, and mitigate discrimination and bias in ML models. IBM also developed AI Explainability 360⁷ to help explain/interpret ML models and AI FactSheets 360⁸ to improve transparency by collecting relevant information (facts) such as the criticality of the model, metrics about dataset, model, service or actions taken during the deployment of the model/service.

Other tools such as Fairlearn [88], What-If [89], Tensor-Flow (TF) Fairness [90], LinkedIn Fairness [91], ML-fairnessgym [92], and UnBias Fairness [93] allow one to check and improve the fairness of ML models. Audit AI [94], TF Model Analyzer/Remediation [95], and REVISE [96] tools also help to build unbiased ML solutions. Tools like SHAP [97], LIME [98], InterpretML [99], and PAIR Facets [100] help to explain and interpret data and ML models. Tools such as Adversarial Robustness Toolbox (ART) [101], PySyft [102], and TF Privacy [103] help secure models and data during ML lifecycle. Data Version Control (DVC) [104] allows one to track and control the versioning of data and models (similar to Git for source code management).

^{6.} https://aif360.mybluemix.net/

^{7.} https://aix360.mybluemix.net/

^{8.} https://aifs360.mybluemix.net/

TABLE 13: AI ethics softwares

| Name | Description |
|---------------------------|--|
| Aeguitas | Bias & Fairness Audit Toolkit |
| Deon | Checklist for data science projects |
| eXplainability Toolbox | Algorithmic bias and explainability |
| | evaluating Fairness Accountability |
| FAT Forensics | and Transparency of AI systems |
| | An Open Source Toolkit for detection |
| AI Fairness 360 | and mitigation of bias in ML models |
| | An Open Source Toolkit for |
| AI Explainability 360 | interpretability and explainability of |
| | datasets and ML models |
| | Trust in AI by increasing trans- |
| AI FactSheets 360 | parency and enabling governance |
| | Toolkit for assessing and improving |
| Fairlearn | fairness in AI |
| Algofairness | BlackBox Auditing Tool |
| FairSight | Fair Decision Making Tool |
| | Analyzing ML datasets |
| PAIR Facets | by visualization |
| What-if Tool | Playing with AI Fairness |
| | A Tool for Exploring Long-Term |
| ML-fairness-gym | Impacts of ML Systems |
| InterpretML | Explain blackbox ML |
| Interpretitie | Measurement of fairness in large |
| LinkedIn Fairness Toolkit | scale ML workflows |
| | Detect demographic differences in |
| Audit Al | the output of ML models |
| SHAP | SHapley Additive exPlanations |
| LD (T | Local Interpretable Model-agnostic |
| LIME | Explanations |
| | Easy computation of fairness metrics |
| IF Fairness Indicators | for binary and multi-class classifiers |
| TF Model Analyzer | Performing model evaluation |
| | Reduces or eliminates user |
| TF Model Remediation | harm resulting from underlying |
| | performance biases |
| Skater | Model Interpretation |
| UnBias Fairness Toolkit | Critical and civic thinking tool |
| DEVICE | A Tool for Measuring and Mitigating |
| REVISE | Bias in Visual Datasets |
| ART | Adversarial Robustness Toolbox |
| | ML Privacy using Federated |
| PySyft | Learning, Differential Privacy, and |
| | Encrypted Computation |
| TF Privacy | Training ML models with |
| | Privacy for training data |
| DVC | Versioning and control of |
| | data and models |

5.2.6 AI Ethics Awareness

Table 14 shows 10 ethical AI awareness training found in the literature, classified by name and organization. Several programs are led by universities and companies to train practitioners, researchers, and citizens on ethical AI principles, guidelines, and applications. The University of Montreal offers a Massive Open Online Course (MOOC) on *Bias and Discrimination in AI* [105]: types of bias, fairness metrics, public laws and policies, technical/institutional bias and discrimination mitigation. The University of Helsinki also provides a course on Ethics of AI; it defines ethics and its principles such as non-maleficence, accountability, transparency, human rights, and fairness.

The Seattle University course on *AI ethics for business* [106] instructs students on ethical issues around the AI technologies and then provides possible mitigations to them. The course *Data Ethics, AI and Responsible Innovation* [107] is also given by the University of Edinburgh. It helps to understand societal/political/legal/ethical issues

| Name | Organization | |
|--------------------------------------|-------------------------|--|
| Data Science Ethics | University of Michigan | |
| Secure & Private AI | Udacity | |
| Ethics in AI and Data Science | Linux Foundation | |
| Ethics of AI | University of Helsinki | |
| AI Ethics for Business | Seattle University | |
| Bias and Discrimination in AI | Université de Montréal | |
| Practical Data Ethics | Fast AI | |
| Data Ethics, AI and | University of Edinburgh | |
| Responsible Innovation | | |
| Ethics for Engineers: Artificial | MIT | |
| Intelligence | | |
| Explainable AI: Scene Classification | Coursera | |
| and GradCam Visualization | | |

around data lifecycle and ethical concepts (e.g., responsibility, data governance) and it also helps to identify, assess, evaluate ethics issues in data science and industry. Massachusetts Institute of Technology (MIT) also provides a course titled *Ethics for Engineers: Artificial Intelligence* [108]. The course help engineers to understand ethics in different cases such as AI, Social media and data, Engineering and society broadly, Engineering and politics broadly, Engineering and democracy. Other courses are provided by training companies such as Coursera, Udacity, and Fast AI to help implement ethical AI into AI projects (e.g., secure and private AI, explainable AI).

5.3 What are the gaps between ethical AI principles and their implementations as well as their root causes?

After identifying current AI ethics implementations, we found several gaps between ethical AI principles and their implementations [66]. In this section, we report about 5 gaps identified between ethical AI principles and their implementations, as well as their root causes (RQ3).

5.3.1 Principle-Implementation Gaps

From the analysis of the literature, we have identified the following gaps:

5.3.1.1 Lack of implementation tools for some ethical AI principles: Several open source tools for AI ethics have been developed including those in Table 13. However, most tools were focused on few principles such as Fairness, Non-discrimination, and Bias (see Table 13). Few tools targeted Explainability, Interpretability, Security, and Privacy principles (e.g., AI Explainability 360, LIME, ART, TF Privacy). In addition, the existing tools do not even cover all aspects of the principles targeted; for example, the fairness implementation is limited to input data, AI models, and outputs without taking into account elements such as the fairness of the AI design/engineering process, the fairness of the AI libraries/tools and infrastructure used in the process, and EthicsOps (i.e., the continuous integration and monitoring of ethics principles in the AI project lifecycle for harm prevention). On the 33 principles, there are missing tools for several principles including Accountability, Responsibility, Traceability, Controllability, Equity, Justice, Constestability, Reliability, Autonomy, Sustainability, Governance, Integrity, Trustworthy, Transparency (partially covered by explainability/interpretability but no traceability), and Inclusiveness.

Summary 4

- The existing ethical guidelines promote education and awareness to foster an ethical mindset, the use of fairness and non-discrimination in products and models as well as data quality control, diversity and inclusiveness of team roles, explainability of model functions and outputs, traceability of all training and operating data.
- The current codes of conduct include laws such as Canada (e.g., Privacy Act), United States (e.g., Privacy Protection Act), Europe (e.g., GDPR), Australia (e.g., Privacy Act), China (e.g., Personal Information Protection Law), and South Africa (e.g., Protection of Personal Information Act); and standards such as Bristish Standards Institution (e.g., BS 8611:2016), CIO Strategy Council (e.g., CAN/CIOSC 101:2019), IEEE (e.g., IEEE 7001-2021), ISO/IEC (e.g., ISO/IEC TR 24028:2020), and NIST (e.g., NISTIR 8332).
- The existing ethical governance frameworks promote the development of inclusive ethical AI governance, strong collaboration and coordination between stakeholder groups and policy makers, the use of algorithmic impact assessment-based solutions for accountability of algorithmic systems.
- The current ethical assessments include checklists such as Deon, Microsoft AI Fairness, Center for Research Libraries [72], and EARN [73]; and algorithmic assessments such as Open Roboethics Institute [74], Data Society [81], UK Statistics Authority [83], European Commission HLEG AI [82], and UNICEF [80].
- The existing ethics tools include Aequitas, Deon, SHAP, LIME, AI FactSheets 360, AI Fairness 360, AI Explainability 360, PAIR Facets, What-If Tool, InterpretML, PySyft, Adversarial Robustness Toolbox, Data Version Control, Tensorflow Privacy.
- The list of institutions currently providing ethical education and awareness materials include : Universite de Montreal (course: *Bias and Discrimination in AI*), Seattle University (course: *AI ethics for business*), University of Edinburgh (course: *Data Ethics, AI and Responsible Innovation*), and MIT (course: *Ethics for Engineers: Artificial Intelligence*)

5.3.1.2 Lack of effective standards: There are few ethical AI standards [51] that cover ethical principles and some are still at the stage of draft [109], [110]. In the studied ethical AI implementations, most standards address AI ethics in terms of governance policies with limited work on technical policies, i.e., step-by-step instructions to apply ethical principles (e.g., transparency, traceability, accountability) during the AI project's lifecycle (i.e., planning, implementation, delivery, and monitoring).

5.3.1.3 Lack of practical training courses on AI ethics: Universities and training companies proposed generic training courses on *what* are AI ethics, guidelines, standards, and very few on *how* AI ethics is (and should be)

really applied during the design and development process of an AI system. There is a need for consistent and up-todate courses on the application of AI ethics in practice.

5.3.1.4 Weakness of the implementation of AI ethics principles in corporate governance: Corporate governance of AI have issues such as improper supervision and monitoring, deficient auditing of financial reports, decision making process issues (e.g., conflicts, divergence), and flaws in multilateral contracts. After analyzing the existing ethical AI governance, very few documents addressed the application of ethical principles to fix these issues, precisely in terms of transparency (e.g., in the auditing of documentation and financial reports, effective application of ethics guidelines and laws), responsibility (e.g., social duty, organisational), equity (e.g., in the decision-making processes with both shareholders and stakeholders, employees and managers), autonomy (e.g., free will of stakeholders), and sustainability (e.g., continuous adaptation of products/services to the market).

5.3.1.5 lack of coverage of ethics in artificial general intelligence by implementation materials: Like T. Hagendorff [6], we found that ethics in Artificial General Intelligence (AGI) and its consequences on humanity are not discussed in implementations materials, while most ethics AI principles cover broad aspects of AGI. It is understood that we are currently at the stage of artificial narrow intelligence (weak AI) and have not yet reached artificial super intelligence [111], but AI is maturing fast and there is a need to anticipate potential ethical issues and consequences [112] that AI may pose in the near future (e.g., controllability, transhumanity).

5.3.2 Root causes of principle-implementation gaps

After analysing the implementations and gaps, the following causes of gaps have been identified.

5.3.2.1 **Human bias**: The studied ethical AI implementations does not provide clear translation from the abstract ethical principles (shared by humans and the society) to ML algorithms and tools. The translation is distorted because (1) most people involved in the implementations are technical (i.e., with a computer science background) and their individual perception [113] limits the implementation, (2) there is lacking proethically designed human-computer interaction or networks of AI systems [12], and (3) it is hard to translate complex human (ethical) values into design tools that are easy to use; which makes them hardly actionable [114].

5.3.2.2 **Poor diversity within the AI community**: The diversity problem is still present in AI research and policy-making teams: (1) all races are not represented in those deciding AI groups (i.e., mainly consisting of white men), (2) there is no equity about genders in the deciding groups (i.e., mainly men and only few women), (3) no inclusion of people with disabilities, and (4) no skill balance (i.e., mainly technical people with science backgrounds). This lack of diversity makes ethical AI implementations (e.g., guidelines, laws, standards, tools) unaligned with ethical AI principles.

5.3.2.3 **Complexity, interconnection of decisions, and processes that are learned from data**: These implementation losses [64] were addressed by the European Parliament. Complexity expresses the behavior of common ML algorithms. For example, ML algorithm modules execute different tasks and the combination provides final results. However, each task has different meanings for data and the randomness of ML algorithms make a step-by-step explanation more challenging [64]. In addition, decisions are often the result of multiple optimization (a batch of simultaneous solutions), that make explainability and interpretability difficult. Furthermore, the ML models that learnt form data are not understandable and transparent. In the context of online learning, the models are continuously updated with data; thus, the application of ethical AI principles must be continuous, or monitored to be effective. In addition, assigning responsibility to ML algorithm harms is difficult and this can lead to issues with moral responsibility [12], [114].

Summary 5

- The 5 identified gaps are : the lack of implementation tools for some ethical principles (i.e., many ethics principles including Accountability/Traceability/Controllability are not supported, no EthicsOps), the lack of effective standards (i.e., limited work on the step-by-step application of ethics principles in the AI lifecycle), the lack of practical training (i.e., lack of practical courses on AI ethics), the weakness of Ethics in corporate governance (e.g., lack of transparency in the documentation and auditing of financial reports, lack of equity in decision making processes, lack of sustainability for services and products), and lack of coverage of ethics in artificial general intelligence by *implementation materials* (i.e., no discussion about this topic and its consequences [112]).
- The 5 identified root causes are *human* (i.e., individual perception bias [113], team selection bias, no proethically designed human-computer interaction [12], translation complexity of ethical values [114]), *diversity* (i.e., no balance about races/genders, no disability inclusion), *complexity, interconnection of decisions, and processes learned from data* (i.e., step-by-step explanation is hard due to the randomness of ML algorithms, multiple optimizations, and hiddenness of ML models), *public-private partnerships and industry-funded research* (i.e., AI democratization issue, AI research freedom [6]), and *lack of metrics* for assessing the implementation of AI ethics principles [115], [116].

5.3.2.4 Issue in public–private partnerships and industry-funded research, in the field of AI: In the studied ethical AI implementations, there are several actors such as inter-governmental partnerships (e.g., OECD, G20), international partnerships (e.g., GPAI, IEEE), national (e.g., Montreal Declaration for a Responsible AI), and universities (e.g., Universite de Montreal, Eindhoven University, ETH Zurich). Most universities are funded by public-private partnerships that limits their freedom in AI research [6] and the democratization of AI; since some results of the research are often proprietary and the research is more focused on the needs of funding partners than on solving real problems in AI. Consequently, these issues affect the implementation of models and support tools for AI ethics.

5.3.2.5 Lack of metrics for ethical AI principles: There is missing proper models and metrics to evaluate the application of those principles [115], [116]. Few metrics have been proposed including fairness [117], privacy [118], and robustness [119] but they are not enough and does not capture well the abstraction of ethical principles. In addition, some work such as [120], [121] tried to provide models for ML traceability but they are not yet effective.

6 GAP MITIGATION

In this section, we provide recommendations to mitigate the identified gaps (RQ4).

Recommandation 1. Several actors (from different genders, races, disabilities, skills) such as policy markers, customers, philosophists, compliance specialists, security experts, software architects and developers must be involved in the design and implementation of the ethics tools, not only technical people (i.e., security experts, software architects, developers), to ensure an effective implementation and application of ethical principles. Some researches have been done to define various metrics for fairness [117], [122], discrimination and bias [122], privacy [118], [123]–[127], security [118], [125], [128], [129], robustness [119], [130], [131], [131]–[135], explainability [136]–[138], interpretability [136]– [138], transparency and accountability [136]-[139], traceability and auditability [140]–[142]. However, more metrics need to be developed to evaluate whether AI systems are ethically-aligned.

Recommandation 2. Teams must aligned projects with EthicsOps to ensure that solutions built follow ethics principles. EthicsOps ensures the continuous application and control of ethics principles during the project lifecycle (planning, execution, delivery). During the planning phase, business requirement specifications must follow existing ethics laws, standards, and guidelines (see Section 5.2). The internal ethics policies of the organization must be updated continuously, since ethical standards and guidelines evolve, as more research results are produced (e.g., models, techniques) in the AI ethics field. In addition, team selection and change management must take into account diversity, inclusiveness, non-discrimination, and equity principles (i.e., equity irrespective of the gender, race, and disability).

Ethical controls using checklists, Algorithmic Impact Assessment (AIA), Key Ethics Indicators (KEI) based on ethics metrics [117], [119], [122]–[125], [128], [130], [136], [137], [140] must be also enforced during the planning, implementation, and delivery of AI projects. During the execution and delivery phases, ethics by design practices [8] help to proactively (by using the principles as requirements) prevent ethical issues from arising. For example, Transparency by design and Fairness by design ensure that outcomes of the project and the project itself are transparent, fair, and auditable by independent third parties (e.g., XAI from EthicalML). In an IT project, EthicsOps can be coupled with existing practices (e.g., DevSecOps [143], AIOps [144]) to design, build, deploy, and monitor ethically-aligned AI systems.

Recommandation 3. More partnerships between publicprivate sectors and universities must be created to accelerate research, development, and knowledge transfer in AI ethics. For example, Facebook (now Meta) partners with the Technical University of Munich (TUM) to support the creation of an independent AI ethics research center with an investment of 7.5 millions USD [145]. In Canada, the Quebec and federal governments invested together an amount of 15 millions CAD in total for the creation of the International Centre of Expertise in Montreal on Artificial Intelligence (CEIMIA)⁹ to foster research and innovation on responsible AI based on ethical principles, human rights, inclusion, diversity, innovation, and economic growth. In addition, research agreements between universities and companies must be clear about time investment (e.g., 50/50), confidentiality, ownership, commercialization rights, retained rights, compliance with laws, publications, warranties, indemnification and liability, and termination.

Summary 6

To mitigate the gap between principles and implementations, the solutions include:

- Inclusiveness and diversity of team roles (races, genders, disabilities, skills).
- Education and awareness on ethical values, culture, methods, and practices [146] to foster an ethical mindset (see Summary 4).
- Increase of partnerships between private-public sectors and universities to accelerate research on AI ethics.
- Application of existing ethics laws and standards during the AI governance and engineering process while staying up-to-date with new changes (see Summary 4).
- Application of EthicsOps in combination with existing practices (e.g., DevSecOps [143], AIOps [144]) to design, build, deploy, and monitor ethically-aligned AI systems from staging to the production environment.

Recommandation 4. More programs on human-centered research, and AI ethics must be included in universities and engineering schools, to instill and foster the adoption of ethical values, culture, methods, and practices [146] by future scientists and engineers. For example, universities such as the University of Montreal, Massachusetts Institute of Technology, Seattle University, and University of Michigan proposed some courses on AI ethics (see Table 14). Scientists and engineers need to know AI ethics in practice and work in an inclusive/diverse environment to build effective and sustainable tools to prevent AI systems from harming. Existing Standards (presented in Summary 4) must be followed during the AI engineering process and these standards must be continuously updated. National and international organizations need to stay abreast of new changes (i.e., new regulatory norms, standard updates).

7 THREATS TO VALIDITY

In this work, we have analyzed 350 reports and websites, and retained 100 reports containing AI principles and AI implementations. The 250 remaining that did not contained AI principles or implementations were ignored. This process was done manually and we may have missed information containing AI principles or implementations.

In addition, the extraction of principles and implementations was done manually with basic functions of Microsoft Excel. It took 2 months to search, read, identify, extract, and record relevant information from the 100 reports and websites. Nevertheless, we have verified document sources and the online publishers to ensure that they are trustworthy. During the extraction process, we have repeated statistics 2 times to avoid any miscalculation.

8 CONCLUSION

In this work, we have analyzed ethical AI principles and their implementations as well as the principleimplementation gaps. We have also proposed some recommendations to mitigate the gaps. Results show that Transparency is the most cited principle among the studied countries. The 11 global principles that cover most continents with a high number of occurrences are Privacy, Transparency, Fairness, Security, Safety, Responsibility, Accountability, Explainability, Well-being, Human Rights, and Inclusiveness. The ethical AI principles published by GPAI, OECD, and G20 are missing the Responsibility principle, compared to the global principles. United States, United Kingdom, and Canada released the highest number of principles between 2016 and 2021. After reviewing more than 121 ethical implementations, 5 principle-implementation gaps were identified including lack of effective implementation tools, lack of effective standards, and weakness of the implementation of AI ethics principles in corporate governance. In addition, 5 root causes of gaps were also found including human bias, lack of metrics for assessing the implementation of ethical principles, poor diversity and inclusion. Recommendations for gap mitigation include: inclusiveness and diversity of team roles, partnerships between public-private sectors and universities for fostering AI ethics research, and the use of EthicsOps. In the future, we plan to develop some ethical tools and propose practical assessment mechanisms, to bridge the gap between ethical AI principles and their implementations.

ACKNOWLEDGMENT

This work is partly funded by the Fonds de Recherche du Québec (FRQ), Natural Sciences and Engineering Research Council of Canada (NSERC), Canadian Institute for Advanced Research (CIFAR), and Mathematics of Information Technology and Complex Systems (MITACS).

REFERENCES

 K. Kourou, T. P. Exarchos, K. P. Exarchos, M. V. Karamouzis, and D. I. Fotiadis, "Machine learning applications in cancer prognosis and prediction," <u>Computational and structural biotechnology</u> journal, vol. 13, pp. 8–17, 2015.

- [2] S. Kuutti, R. Bowden, Y. Jin, P. Barber, and S. Fallah, "A survey of deep learning applications to autonomous vehicle control," <u>IEEE</u> <u>Transactions on Intelligent Transportation Systems</u>, vol. 22, no. 2, pp. 712–733, 2020.
- [3] G. Gui, F. Liu, J. Sun, J. Yang, Z. Zhou, and D. Zhao, "Flight delay prediction based on aviation big data and machine learning," <u>IEEE Transactions on Vehicular Technology</u>, vol. 69, no. 1, pp. <u>140–150</u>, 2020.
- [4] L. Floridi, J. Cowls, M. Beltrametti, R. Chatila, P. Chazerand, V. Dignum, C. Luetge, R. Madelin, U. Pagallo, F. Rossi et al., "Ai4people—an ethical framework for a good ai society: Opportunities, risks, principles, and recommendations," <u>Minds and</u> Machines, vol. 28, no. 4, pp. 689–707, 2018.
- [5] A. Jobin, M. Ienca, and E. Vayena, "The global landscape of ai ethics guidelines," <u>Nature Machine Intelligence</u>, vol. 1, no. 9, pp. 389–399, 2019.
- [6] T. Hagendorff, "The ethics of ai ethics: An evaluation of guidelines," Minds and Machines, vol. 30, no. 1, pp. 99–120, 2020.
- [7] J. Fjeld, N. Achten, H. Hilligoss, A. Nagy, and M. Srikumar, "Principled artificial intelligence: Mapping consensus in ethical and rights-based approaches to principles for ai," <u>Berkman Klein</u> Center Research Publication, no. 2020-1, 2020.
- [8] E. Commission, "Ethics by design and ethics of approaches for artificial intelligence," Tech. use https://ec.europa.eu/info/funding-Rep., 2021. tenders/opportunities/docs/2021-2027/horizon/guidance/ethicsby-design-and-ethics-of-use-approaches-for-artificialintelligence_he_en.pdf.
- [9] E. Shearer, R. Stirling, and W. Pasquarelli, "Government ai readiness index," Tech. Rep., 2020, https://static1.squarespace.com/static/58b2e92c1e5b6c828058484 e/t/5f7747f29ca3c20ecb598f7c/1601653137399/AI+Readiness+ Report.pdf.
- [10] P. F. Nettel, A. Rogerson, T. Westgarth, K. Iida, H. Mbayo, A. Finotto, S. Rahim, and A. Petheram, "Government ai readiness index," Tech. Rep., 2021, https://static1.squarespace.com/static/58b2e92c1e5b6c828058484 e/t/61ead0752e7529590e98d35f/1642778757117/Government_AI _Readiness_21.pdf.
- [11] "Aiethics," 2022, https://github.com/ngatilio/AIEthics.
- [12] J. Morley, L. Floridi, L. Kinsey, and A. Elhalal, "From what to how: an initial review of publicly available ai ethics tools, methods and research to translate principles into practices," in <u>Ethics</u>, <u>Governance</u>, and <u>Policies in Artificial Intelligence</u>. Springer, 2021, pp. 153–183.
- [13] B. Goehring, F. Rossi, and D. Zaharchuk, "Advancing ai ethics beyond compliance," IBM Corporation, April, 2020.
- [14] A. Kaminski, T. Krafft, M. Hauer, C. Hustedt, L. Fetic, and C. Hubig, "From principles to practice. an interdisciplinary framework to operationalise ai ethics." 2020.
- [15] D. Leslie, "Understanding artificial intelligence ethics and safety: A guide for the responsible design and implementation of ai systems in the public sector," <u>Available at SSRN 3403301</u>, 2019.
 [16] A. H. Nelson, D. Jenkins, S. Zanti, M. F. Katz, E. Berkowitz,
- [16] A. H. Nelson, D. Jenkins, S. Zanti, M. F. Katz, E. Berkowitz, T. Burnett, and D. P. Culhane, <u>A toolkit for centering racial equity</u> throughout data integration. <u>Actionable Intelligence for Social</u> <u>Policy, University of Pennsylvania, 2020.</u>
- [17] N. Smuha, "Ethics guidelines for trustworthy ai," in AI & Ethics, Date: 2019/05/28-2019/05/28, Location: Brussels (Digityser), Belgium, 2019.
- [18] T. F. for Best Practices in Machine Learning, "Technical and organizational best practices," Tech. Rep., 2021, https://www.nist.gov/system/files/documents/2021/08/18/airmf-rfi-0010-attachment3.pdf.
- [19] W. E. Forum, "Guidelines for ai procurement," Tech. Rep., 2019, https://www3.weforum.org/docs/WEF_Guidelines_for_AI_Pro curement.pdf.
- [20] "Ethically aligned design," Tech. Rep., 2020, https://standards.ieee.org/wpcontent/uploads/import/documents/other/ ead_v2.pdf.
- [21] OCDE, Tools and Ethics for Applied Behavioural Insights: The BASIC Toolkit, 2019. [Online]. Available: https://www.oecdilibrary.org/content/publication/9ea76a8f-en
- [22] I. A. of Privacy Professionals, "Building ethics into privacy frameworks for big data and ai," Tech. Rep., 2018, https://iapp.org/media/pdf/resource_center/building_ethics_ into_privacy_frameworks_white_paper.pdf.

- [23] U. Pagallo, P. Aurucci, P. Casanovas, R. Chatila, P. Chazerand, V. Dignum, C. Luetge, R. Madelin, B. Schafer, and P. Valcke, "On good ai governance: 14 priority actions, a smart model of governance, and a regulatory toolbox," 2019.
- [24] S. Taylor, M. Boniface, B. Pickering, M. Anderson, D. Danks, A. Følstad, M. Leese, V. Müller, T. Sorell, A. Winfield et al., "Responsible ai–key themes, concerns & recommendations for european research and innovation," 2018.
- [25] U. I. C. Office, "Guidance on the ai auditing framework - draft guidance for consultation," 2020, https://ico.org.uk/media/2617219/guidance-on-the-aiauditing-framework-draft-for-consultation.pdf.
- [26] I. C. Office, "Guidance on ai and data protection," 2020, https://ico.org.uk/media/for-organisations/guide-to-dataprotection/key-dp-themes/guidance-on-ai-and-data-protection-0-0.pdf.
- [27] J. L. Leidner and V. Plachouras, "Ethical by design: Ethics best practices for natural language processing," in <u>Proceedings of the First ACL Workshop on Ethics in Natural Language Processing</u>, 2017, pp. 30–40.
- [28] S. Vallor, "An ethical toolkit for engineerpractice," ing/design Markkula Center for Applied Santa Clara https://www. Ethics, University, scu. edu/ethics-in-technology-practice/ethical-toolkit, 2018.
- [29] K. Zhou, "Design ethically toolkit," 2018. [Online]. Available: https://www.designethically.com/toolkit
- [30] "Ethics canvas," 2016. [Online]. Available: https://opengogs.adaptcentre.ie/ADAPT/ethics-canvas
- [31] "Privacy act," 1985. [Online]. Available: https://lawslois.justice.gc.ca/eng/acts/p-21/fulltext.html
- [32] "Personal information protection and electronic documents act," 2000. [Online]. Available: https://lawslois.justice.gc.ca/ENG/ACTS/P-8.6/index.html
- [33] "Financial consumer agency of canada act," 2001. [Online]. Available: https://laws-lois.justice.gc.ca/eng/acts/F-11.1/
- [34] "Health insurance portability and accountability act of 1996," 1996. [Online]. Available: https://www.hhs.gov/hipaa/forprofessionals/index.html
- [35] "Privacy act," 1974. [Online]. Available: https://lawslois.justice.gc.ca/eng/acts/F-11.1/
- [36] "Privacy protection act," 1980. [Online]. Available: https://lawslois.justice.gc.ca/eng/acts/F-11.1/
- [37] "California consumer privacy act," 2018. [Online]. Available: https://leginfo.legislature.ca.gov/faces/billTextClient.xhtml?bill _id=201720180SB1121
- [38] "Eu-u.s. and swiss-u.s. privacy shield," 2016. [Online]. Available: https://www.privacyshield.gov/program-overview
- [39] "Fair credit reporting act," 2018. [Online]. Available: https://www.ftc.gov/legal-library/browse/statutes/faircredit-reporting-act
- [40] "Gramm-leach-bliley act," 1999. [Online]. Available: https://www.govinfo.gov/content/pkg/PLAW-106publ102/pdf/PLAW-106publ102.pdf
- [41] "Data protection act," 2018. [Online]. Available: https://www.hhs.gov/hipaa/for-professionals/index.html
- [42] "General data protection regulation," 2016. [Online]. Available: https://gdpr-info.eu/
- [43] "Privacy act," 1988. [Online]. Available: https://www.oaic.gov.au/privacy/the-privacy-act
- [44] "Personal information protection law," 2021. [Online]. Available: https://www.china-briefing.com/news/the-prc-personalinformation-protection-law-final-a-full-translation/
- [45] "Cyber security law," 2016. [Online]. Available: https://www.tradecommissioner.gc.ca/china-chine/cybersecurity_cyber-securite_china-chine.aspx?lang=eng
- [46] "Personal data protection act," 2012. [Online]. Available: https://sso.agc.gov.sg/Act/PDPA2012
- [47] "Protection from online falsehoods and manipulation act," 2019.[Online]. Available: https://sso.agc.gov.sg/Acts-Supp/18-2019
- [48] "Protection of personal information act," 2013. [Online]. Available: https://www.gov.za/sites/default/files/gcis_document/201409 /3706726-11act4of2013popi.pdf
- [49] "Data protection law," 2021. [Online]. Available: https://u.ae/en/about-the-uae/digital-uae/data/dataprotection-laws

- [50] "Bundesdatenschutzgesetz," 2009. [Online]. Available: https://www.datenschutz-wiki.de/BDSG_2009
- [51] "Oceanis." [Online]. Available: https://ethicsstandards.org/repository/
- [52] B. BS8611, "Robots and robotic devices, guide to the ethical design and application of robots and robotic systems," <u>British</u> Standards Institute, 2016.
- [53] "Ethical design and use of automated decision systems," 2019. [Online]. Available: https://viewer.joomag.com/ciosc-standardcan-ciosc-101-2019/0672323001570024164
- [54] "Ieee guide for the use of artificial intelligence exchange and service tie to all test environments (ai-estate)," IEEE Std 1232.3-2014, pp. 1–102, 2014.
 [55] "Information technology artificial intelligence overview
- [55] "Information technology artificial intelligence overview of trustworthiness in artificial intelligence," 2020. [Online]. Available: https://www.iso.org/standard/77608.html
- [56] "Artificial intelligence," 2017. [Online]. Available: https://www.iso.org/committee/6794475.html
- [57] "Ieee standard model process for addressing ethical concerns during system design," <u>IEEE Std 7000-2021</u>, pp. 1–82, 2021.
- [58] A. F. Winfield, S. Booth, L. A. Dennis, T. Egawa, H. Hastie, N. Jacobs, R. I. Muttram, J. I. Olszewska, F. Rajabiyazdi, A. Theodorou et al., "Ieee p7001: a proposed standard on transparency," Frontiers in Robotics and AI, p. 225, 2021.
- [59] "Ieee ontological standard for ethically driven robotics and automation systems," IEEE Std 7007-2021, pp. 1–119, 2021.
- [60] B. Stanton, T. Jensen <u>et al.</u>, "Trust and artificial intelligence," <u>preprint</u>, 2021.
 [61] "Acm code of ethics and profes-
- [61] "Acm code of ethics and professional conduct," 2018. [Online]. Available: https://www.acm.org/binaries/content/assets/about/acmcode-of-ethics-booklet.pdf
- [62] V. Dessain, O. Meier, and V. Salas, "Corporate governance and ethics: Shareholder reality, social responsibility or institutional necessity?" M@ n@ gement, vol. 11, no. 2, pp. 65–79, 2008.
- [63] "Ethics and governance of ai," 2022, https://cyber.harvard.edu/topics/ethics-and-governance-ai.
- [64] A. Koene, C. Clifton, Y. Hatada, H. Webb, and R. Richardson, "A governance framework for algorithmic accountability and transparency," 2019.
- [65] A. Dafoe, "Ai governance: a research agenda," <u>Governance of AI Program, Future of Humanity Institute, University of Oxford:</u> Oxford, UK, vol. 1442, p. 1443, 2018.
- [66] C. Lütge, "White paper ai ethics and governance," Tech. Rep., 2020, https://ieai.mcts.tum.de/wpcontent/uploads/2020/04/White-Paper_AI-Ethics-and-Governance-_March-20201.pdf.
- [67] A. F. Winfield, K. Michael, J. Pitt, and V. Evers, "Machine ethics: The design and governance of ethical ai and autonomous systems [scanning the issue]," <u>Proceedings of the IEEE</u>, vol. 107, no. 3, pp. 509–517, 2019.
- [68] "Deon : An ethics checklist for data scientists," 2016. [Online]. Available: deon.drivendata.org/
- [69] C. S. E. Institute, "Designing ethical ai experiences: Checklist and agreement," Tech. Rep., 2019, https://resources.sei.cmu.edu/asset_files/FactSheet/2019_010_0 01_636622.pdf.
- [70] "Ai fairness checklist," 2016. [Online]. Available: https://www.microsoft.com/en-us/research/project/aifairness-checklist/
- [71] AlgorithmWatch, "Automated decision-making systems in the public sector: An impact assessment tool for public authorities," Tech. Rep., 2021, https://algorithmwatch.org/en/wpcontent/uploads/2021/09/2021_AW_Decision_Public_Sector_EN _v5.pdf.
- [72] T. C. for Research Libraries, "Trustworthy repositories audit & certification: Criteria and checklist," Tech. Rep., 2007, https://www.crl.edu/sites/default/files/d6/attachments/pages /trac_0.pdf.
- [73] T. E. Assistance and R. N. on Disability Inclusion, "Checklist for employers: Facilitating the hiring of people with disabilities through the use of erecruiting screening systems, including ai," Tech. Rep., 2020, https://production-askearnorg.s3.amazonaws.com/EARN_PEAT_e_Recruiting_Checklist456 e322.pdf.
- [74] O. R. Institute, "Foresight into ai ethics (faie): A toolkit for creating an ethics roadmap for your ai

project," Tech. Rep., 2019, https://openroboethics.org/wpcontent/uploads/2021/07/ORI-Foresight-into-Artificial-Intelligence-Ethics-Launch-V1.pdf.

- [75] "Algorithmic impact assessment," 2022, https://canadaca.github.io/aia-eia-js/.
- [76] E. P. for the Information Society, "Artificial intelligence impact assessment," Tech. Rep., 2018, https://static1.squarespace.com/static/5b7877457c9327fa97fef427 /t/5c368c611ae6cf01ea0fba53/1547078768062/Artificial+Intelli gence+Impact+Assessment+-+English.pdf.
- [77] D. Reisman, J. Schultz, K. Crawford, and M. Whittaker, "Algorithmic impact assessments: A practical framework for public agency accountability," Tech. Rep., 2018, https://ainowinstitute.org/aiareport2018.pdf.
- [78] "Algorithmic equity toolkit," Tech. Rep., 2020, https://www.aclu-wa.org/docs/algorithmic-equity-toolkitdigital-download.
- [79] "Ai system ethics self-assessment tool," 2020. [Online]. Available: https://www.digitaldubai.ae/self-assessment
- [80] D. for Children Collaborative with UNICEF, "Ethical assessment," Tech. Rep., 2020, https://static1.squarespace.com/static/5ced8e8d4d23c100012bd9 ee/t/5ea15e953b46f95d97703164/1587633826177/Ethics+and+Safe guarding+Compilation+Pack+FINAL+1.pdf.
- [81] E. Moss, E. A. Watkins, R. Singh, M. C. Elish, and J. Metcalf, "Assembling accountability: algorithmic impact assessment for the public interest," <u>Available at SSRN 3877437</u>, 2021.
- [82] "Altai the assessment list trustworthy on artificial intelligence," 2019. [Online]. Available: https://futurium.ec.europa.eu/en/european-aialliance/pages/altai-assessment-list-trustworthy-artificialintelligence
- [83] U. S. Authority, "National statisticians data ethics advisory committee - guidelines on using the ethics self-assessment process," Tech. Rep., 2021, https://uksa.statisticsauthority.gov.uk/wpcontent/uploads/2021/04/2021_Self-assessment_guidance _V2.3.pdf.
- [84] M. E. Kaminski and G. Malgieri, "Algorithmic impact assessments under the gdpr: producing multi-layered explanations," <u>International Data Privacy Law</u>, pp. 19–28, 2020.
- [85] B. Lepri, N. Oliver, E. Letouzé, A. Pentland, and P. Vinck, "Fair, transparent, and accountable algorithmic decision-making processes," <u>Philosophy & Technology</u>, vol. 31, no. 4, pp. 611–627, 2018.
- [86] "Bias and fairness audit toolkit," 2018. [Online]. Available: http://aequitas.dssg.io/
- [87] "Fat forensics: Algorithmic fairness, accountability and transparency toolbox," 2018. [Online]. Available: https://github.com/fat-forensics/fat-forensics
- [88] "Fairlearn." [Online]. Available: https://github.com/fairlearn/fairlearn
- [89] "What-if tool." [Online]. Available: https://github.com/PAIRcode/what-if-tool
- [90] "Fairness indicators." [Online]. Available: https://github.com/tensorflow/fairness-indicators
- [91] "The linkedin fairness toolkit (lift)." [Online]. Available: https://github.com/linkedin/LiFT
- [92] "Ml-fairness-gym." [Online]. Available: https://github.com/google/ml-fairness-gym
- [93] "Unbias fairness toolkit." [Online]. Available: https://unbias.wp.horizon.ac.uk/fairness-toolkit
- [94] "Audit ai." [Online]. Available: https://github.com/pymetrics/audit-ai
- [95] "Tensorflow model remediation." [Online]. Available: https://github.com/tensorflow/model-remediation
- [96] A. Wang, A. Liu, R. Zhang, A. Kleiman, L. Kim, D. Zhao, I. Shirai, A. Narayanan, and O. Russakovsky, "Revise: Revealing visual biases," 2021. [Online]. Available: https://github.com/princetonvisualai/revise-tool
- [97] "Revise: Revealing visual biases." [Online]. Available: https://github.com/slundberg/shap
- [98] "Lime." [Online]. Available: https://github.com/marcotcr/lime
- [99] "Interpretml alpha release." [Online]. Available: https://github.com/interpretml/interpret
- [100] "Facets." [Online]. Available: https://github.com/PAIRcode/facets

- [101] "Adversarial robustness toolbox (art)." [Online]. Available: https://github.com/Trusted-AI/adversarial-robustness-toolbox
 [102] "Pysyft." [Online]. Available:
- https://github.com/OpenMined/PySyft [103] "Tensorflow privacy." [Online]. Available:
- https://github.com/tensorflow/privacy [104] "Data version control." [Online]. Available:
- https://github.com/iterative/dvc
- [105] "Mooc bias and discrimination in ai," 2021. [Online]. Available: https://www.edx.org/course/bias-and-discrimination-in-ai
- [106] "Ethics and transformative technologies," 2020. [Online]. Available: https://www.seattleu.edu/ethics-and-technology/educational-resources/short-course/
- [107] "Data ethics, ai and responsible innovation," 2021. [Online]. Available: https://www.edx.org/course/Data-Ethics-AI-and-Responsible-Innovation
- [108] "Ethics for engineers: Artificial intelligence," 2020. [Online]. Available: https://ocw.mit.edu/courses/10-01-ethics-forengineers-artificial-intelligence-spring-2020/pages/syllabus/
- [109] S. O'Sullivan, N. Nevejans, C. Allen, A. Blyth, S. Leonard, U. Pagallo, K. Holzinger, A. Holzinger, M. I. Sajid, and H. Ashrafian, "Legal, regulatory, and ethical frameworks for development of standards in artificial intelligence (ai) and autonomous robotic surgery," <u>The international journal of medical robotics and computer assisted surgery</u>, vol. 15, no. 1, p. e1968, 2019.
- [110] B. Mittelstadt, "Ai ethics-too principled to fail," <u>arXiv preprint</u> arXiv:1906.06668, 2019.
- [111] R. V. Yampolskiy, Artificial superintelligence: a futuristic approach. cRc Press, 2015.
- [112] N. Bostrom, "Existential risks," Journal of Evolution and technology, vol. 9, no. 1, pp. 1–31, 2002.
 [113] M. Arvan, "Mental time-travel, semantic flexibility, and ai ethics,"
- [113] M. Arvan, "Mental time-travel, semantic flexibility, and ai ethics," AI & SOCIETY, pp. 1–20, 2018.
- [114] V. Vakkuri, K.-K. Kemell, J. Kultanen, M. Siponen, and P. Abrahamsson, "Ethically aligned design of autonomous systems: Industry viewpoint and an empirical study," <u>arXiv preprint</u> arXiv:1906.07946, 2019.
- [115] R. Thomas and D. Uminsky, "Reliance on metrics is a fundamental challenge for ai," in Ethics of Data Science Conference, 2020.
- [116] R. Eitel-Porter, "Beyond the promise: implementing ethical ai," AI and Ethics, vol. 1, no. 1, pp. 73–80, 2021.
- [117] S. Corbett-Davies and S. Goel, "The measure and mismeasure of fairness: A critical review of fair machine learning," <u>arXiv</u> preprint arXiv:1808.00023, 2018.
- [118] P. Mohassel and Y. Zhang, "Secureml: A system for scalable privacy-preserving machine learning," in 2017 IEEE symposium on security and privacy (SP). IEEE, 2017, pp. 19–38.
- [119] M.-I. Nicolae, M. Sinn, M. N. Tran, B. Buesser, A. Rawat, M. Wistuba, V. Zantedeschi, N. Baracaldo, B. Chen, H. Ludwig et al., "Adversarial robustness toolbox v1. 0.0," <u>arXiv preprint</u> arXiv:1807.01069, 2018.
- [120] A. Sablayrolles, M. Douze, C. Schmid, and H. Jégou, "Radioactive data: tracing through training," in <u>International Conference on</u> Machine Learning. PMLR, 2020, pp. 8326–8335.
- [121] M. Mora-Cantallops, S. Sánchez-Alonso, E. García-Barriocanal, and M.-A. Sicilia, "Traceability for trustworthy ai: A review of models and tools," <u>Big Data and Cognitive Computing</u>, vol. 5, no. 2, 2021. [Online]. Available: https://www.mdpi.com/2504-2289/5/2/20
- [122] N. Mehrabi, F. Morstatter, N. Saxena, K. Lerman, and A. Galstyan, "A survey on bias and fairness in machine learning," <u>ACM</u> <u>Comput. Surv.</u>, vol. 54, no. 6, jul 2021.
- [123] M. Abadi, A. Chu, I. Goodfellow, H. B. McMahan, I. Mironov, K. Talwar, and L. Zhang, "Deep learning with differential privacy," in Proceedings of the 2016 ACM SIGSAC conference on computer and communications security, 2016, pp. 308–318.
- [124] L. Sweeney, "k-anonymity: A model for protecting privacy," International Journal of Uncertainty, Fuzziness and Knowledge-Based Systems, vol. 10, no. 05, pp. 557–570, 2002.
- [125] K. Bonawitz, V. Ivanov, B. Kreuter, A. Marcedone, H. B. McMahan, S. Patel, D. Ramage, A. Segal, and K. Seth, "Practical secure aggregation for privacy-preserving machine learning," in proceedings of the 2017 ACM SIGSAC Conference on Computer and Communications Security, 2017, pp. 1175–1191.
- and Communications Security, 2017, pp. 1175–1191.
 [126] R. Mendes and J. P. Vilela, "Privacy-preserving data mining: Methods, metrics, and applications," <u>IEEE Access</u>, vol. 5, pp. 10562–10582, 2017.

- [127] V. S. Iyengar, "Transforming data to satisfy privacy constraints," ser. KDD '02. New York, NY, USA: Association for Computing Machinery, 2002, p. 279–288.
- [128] M. Barreno, B. Nelson, A. D. Joseph, and J. D. Tygar, "The security of machine learning," <u>Machine Learning</u>, vol. 81, no. 2, pp. 121– 148, 2010.
- [129] N. Papernot, P. McDaniel, A. Sinha, and M. Wellman, "Towards the science of security and privacy in machine learning," <u>arXiv</u> preprint arXiv:1611.03814, 2016.
- [130] N. Carlini and D. Wagner, "Towards evaluating the robustness of neural networks," in 2017 IEEE Symposium on Security and Privacy (SP), 2017, pp. 39–57.
 [131] W. Brendel, J. Rauber, and M. Bethge, "Decision-based adversar-
- [131] W. Brendel, J. Rauber, and M. Bethge, "Decision-based adversarial attacks: Reliable attacks against black-box machine learning models," arXiv preprint arXiv:1712.04248, 2017.
- [132] N. Papernot, P. McDaniel, X. Wu, S. Jha, and A. Swami, "Distillation as a defense to adversarial perturbations against deep neural networks," in 2016 IEEE symposium on security and privacy (SP). IEEE, 2016, pp. 582–597.
- [133] S.-M. Moosavi-Dezfooli, A. Fawzi, O. Fawzi, and P. Frossard, "Universal adversarial perturbations," in Proceedings of the IEEE conference on computer vision and pattern recognition, 2017, pp. 1765–1773.
- [134] N. Papernot, F. Faghri, N. Carlini, I. Goodfellow, R. Feinman, A. Kurakin, C. Xie, Y. Sharma, T. Brown, A. Roy et al., "Technical report on the cleverhans v2. 1.0 adversarial examples library," arXiv preprint arXiv:1610.00768, 2016.
- [135] J. Rauber, W. Brendel, and M. Bethge, "Foolbox: A python toolbox to benchmark the robustness of machine learning models," <u>arXiv</u> preprint arXiv:1707.04131, 2017.
- [136] M. T. Ribeiro, S. Singh, and C. Guestrin, "" why should i trust you?" explaining the predictions of any classifier," in Proceedings of the 22nd ACM SIGKDD international conference on knowledge discovery and data mining, 2016, pp. 1135–1144.
- [137] S. M. Lundberg and S.-I. Lee, "A unified approach to interpreting model predictions," Advances in neural information processing systems, vol. 30, 2017.
- [138] D. V. Carvalho, E. M. Pereira, and J. S. Cardoso, "Machine learning interpretability: A survey on methods and metrics," Electronics, vol. 8, no. 8, p. 832, 2019.
- [139] M. Mitchell, S. Wu, A. Zaldivar, P. Barnes, L. Vasserman, B. Hutchinson, E. Spitzer, I. D. Raji, and T. Gebru, "Model cards for model reporting," in Proceedings of the conference on fairness, accountability, and transparency, 2019, pp. 220–229.
- [140] J. Weng, J. Weng, J. Zhang, M. Li, Y. Zhang, and W. Luo, "Deepchain: Auditable and privacy-preserving deep learning with blockchain-based incentive," IEEE Transactions on Dependable and Secure Computing, vol. 18, no. 5, pp. 2438–2455, 2019.
- [141] R.-Y. Chen, "A traceability chain algorithm for artificial neural networks using t-s fuzzy cognitive maps in blockchain," <u>Future</u> <u>Generation Computer Systems</u>, vol. 80, pp. 198–210, 2018.
- [142] M. Mora-Cantallops, S. Sánchez-Alonso, E. García-Barriocanal, and M.-A. Sicilia, "Traceability for trustworthy ai: A review of models and tools," <u>Big Data and Cognitive Computing</u>, vol. 5, no. 2, p. 20, 2021.
- [143] H. Myrbakken and R. Colomo-Palacios, "Devsecops: a multivocal literature review," in International Conference on Software <u>Process Improvement and Capability Determination</u>. Springer, 2017, pp. 17–29.
- [144] Y. Dang, Q. Lin, and P. Huang, "Aiops: real-world challenges and research innovations," in 2019 IEEE/ACM 41st International Conference on Software Engineering: Companion Proceedings (ICSE-Companion). IEEE, 2019, pp. 4–5.
- [145] "Facebook and the technical university of munich announce new independent tum institute for ethics in artificial intelligence," 2019. [Online]. Available: https://about.fb.com/news/2019/01/tum-institute-forethics-in-ai/
- [146] B. Taebi, J. van den Hoven, and S. J. Bird, "The importance of ethics in modern universities of technology," <u>Science and Engineering Ethics</u>, vol. 25, no. 6, pp. 1625–1632, 2019.